# University of Nebraska - Lincoln Digital Commons@University of Nebraska - Lincoln

Sociology Theses, Dissertations, & Student Research

Sociology, Department of

4-2016

# On The Street and On Campus: A Comparison of Life Course Trajectories Among Homeless and College Lesbian, Gay, Bisexual, Transgender and Queer Young Adults

Rachel M. Schmitz University of Nebraska-Lincoln, raschm02@gmail.com

Follow this and additional works at: http://digitalcommons.unl.edu/sociologydiss



Part of the Family, Life Course, and Society Commons, and the Gender and Sexuality Commons

Schmitz, Rachel M., "On The Street and On Campus: A Comparison of Life Course Trajectories Among Homeless and College Lesbian, Gay, Bisexual, Transgender and Queer Young Adults" (2016). Sociology Theses, Dissertations, & Student Research. 40. http://digitalcommons.unl.edu/sociologydiss/40

This Article is brought to you for free and open access by the Sociology, Department of at DigitalCommons@University of Nebraska - Lincoln. It has been accepted for inclusion in Sociology Theses, Dissertations, & Student Research by an authorized administrator of DigitalCommons@University of Nebraska - Lincoln.

# ON THE STREET AND ON CAMPUS: A COMPARISON OF LIFE COURSE TRAJECTORIES AMONG HOMELESS AND COLLEGE LESBIAN, GAY, BISEXUAL, TRANSGENDER AND QUEER YOUNG ADULTS

by

Rachel M. Schmitz

# A DISSERTATION

Presented to the Faculty of

The Graduate College at the University of Nebraska

In Partial Fulfillment of Requirements

For the Degree of Doctor of Philosophy

Major: Sociology

(Women's and Gender Studies)

Under the Supervision of Professor Kimberly A. Tyler

Lincoln, Nebraska

May, 2016

ON THE STREET AND ON CAMPUS: A COMPARISON OF LIFE COURSE

TRAJECTORIES AMONG HOMELESS AND COLLEGE LESBIAN, GAY, BISEXUAL, TRANSGENDER AND QUEER YOUNG ADULTS

Rachel Marie Schmitz, Ph.D.

University of Nebraska, 2016

Advisor: Kimberly A. Tyler

This study examines the life course experiences of lesbian, gay, bisexual, transgender and queer (LGBTQ) homeless young adults and LGBTQ college students. Though both of these groups have in common their age (i.e. young adults) and LGBTQ identity, college students generally have more resources and are expected to fare better into later life compared to homeless young adults. Despite these disparities, all LGBTQ young people are likely at greater risk for negative health outcomes and social issues due to their status as sexual and/or gender minorities. Little research, however, has simultaneously examined these two groups, and how their life course experiences uniquely differ based on social environments (i.e. college vs. homeless). Using in-depth, face-to-face interviews with 46 LGBTQ young adults between 19 and 26 years of age, I examine how homeless young people and college students navigate their sexual and gender identities, social contexts and relationships. Findings reveal the importance of social context in identity development, such that college students largely found the college context to be conducive to identity growth, while homeless young adults viewed homelessness as a hindrance to addressing identity-related issues, as they needed to focus on survival. Furthermore, all LGBTQ young adults strategically managed their identities in distinctive ways depending on the social context and relationship, with college

students' tactics being tied to maintaining their reputations, while homeless young adults' motivations were linked to ensuring their physical safety on the streets. Finally, the majority of LGBTQ young adults conceptualized their identity-related challenges as making them stronger and more resilient by enhancing their social relationships and imbuing them with confidence and empathy. Homeless young adults viewed their challenges in homelessness as more transformative compared to their experiences with sexuality and gender-related prejudice and discrimination. These findings alert service providers and policymakers to the fact that programs need to be tailored to LGBTQ young adults based on their life course experiences. This study also highlights the importance of understanding LGBTQ young adults not as a monolithic social group, but one that is rich with both similarities and distinctions across social context, including the homeless and college environments.

# COPYRIGHT

© 2016, Rachel M. Schmitz

# **ACKNOWLEDGEMENTS**

There are many people to whom I extend my gratitude for their support and guidance in completing this dissertation. First and foremost, I would like to thank my academic advisor and committee chair, Dr. Kimberly A. Tyler, for her unending support and encouragement throughout my sociological career. She has continually provided me with the tools and resources that have helped me establish myself as an independent researcher and academic. Furthermore, my committee members, Dr. Emily Kazyak, Dr. Helen Moore and Dr. Wayne Babchuk, offered invaluable insight into the development and refinement of my dissertation by encouraging me to think beyond my own interpretations, and for this I am truly grateful.

Additionally, I extend my appreciation towards the lesbian, gay, bisexual, transgender and queer young adults who willingly shared their stories and life experiences with me. Without the participation of these college students and homeless young adults, this study would not have been possible. I hope that this research benefits the lives of LGBTQ young adults in a meaningful way through its focus on the unique impact of social context and identity. Relatedly, I am so thankful for the numerous individuals and organizations that aided in the recruitment process, including local homeless service agencies and university faculty and staff.

I also give my thanks to the University of Nebraska-Lincoln Sociology

Department for believing in my potential as a graduate student and supporting my

academic growth and professional development. The Sociology Department provided

funding for this research and this assistance was invaluable in the process of data collection and allowing me to maintain a timely dissertation timeline.

Last but not least, I am eternally grateful for my family and friends who have always supported my life path and have stood by my side throughout this journey as my dream of earning my PhD has reached fruition. I thank you all for your unconditional love and support.

# TABLE OF CONTENTS

LIST OF TABLES	viii
CHAPTER 1	1
CHAPTER 2	6
CHAPTER 3	14
CHAPTER 4	19
CHAPTER 5	27
CHAPTER 6	63
REFERENCES	85
APPENDICES	110
Appendix A: Institutional Review Board Approval Letters	111
Appendix B: Participant Informed Consent Form	114
Appendix C: Study Recruitment Flyers	117
Appendix D: Study Email Recruitment Scripts	120
Appendix E: Study Referral Recruitment Cards	123
Appendix F: List of Services for Participants	126
Appendix G: Screener Eligibility Questions	130
Appendix H: Demographic Questionnaire	132
Appendix I: Interview Grand Tour Questions	135

	0.5		1
List	ot i	ľah	les

T 11 1	n	1 4	D	1 1	ъ.	$\alpha$ · 1 $\alpha$ · ·	2
Table I	: Kes	ponaent	Demogra	pnics by	Primary	Social Context	 20

### CHAPTER 1

# INTRODUCTION

Adolescence and the transition to adulthood tend to be fraught with uncertainty and complexity, at both the individual and structural levels (Arnett 2000). Young people struggle to exert agency within their life course opportunities (Clausen 1991) while social structures, such as families and education, work to shape young adults' beliefs regarding their future lives and relationships (Plotnick 2007). While primary markers of adulthood are perceived as milestones of pathways to growing up, including education, careers and family formation (Shanahan 2000), there is increasing heterogeneity in how people traverse life course trajectories and the ways that these experiences distinctly shape development (George 1993; McLeod and Almazan 2003). The tension between structural constraints and the individual's ability to exert agency over their life course trajectories is uniquely salient in the transition to adulthood as young people reside in a limbo-like status between adolescent development and the more established status of middle life.

Identity, which can be defined as one's sense of self, conceptualized internally and externally through social interactions, cultural contexts and overlapping, intersecting sources of oppression (Jones and McEwan 2000), is key to understanding how young people navigate their lives, More specifically, processes of unstable identity fluctuations, which characterize adolescence and young adulthood, are prominent in young people's establishment of sexual and gender orientations (Savin-Williams and Ream 2007). Approximately 6.4% of young adults aged 18-29 identify as lesbian, gay, bisexual or transgender (LGBT) (Gates and Newport 2012). As young people develop their sexual selves, a sense of family connectedness and support is crucial for maintaining positive

mental health, especially when youth do not subscribe to the norm of heterosexuality and fall somewhere in the LGBT spectrum (Needham and Austin 2010). The majority of studies, however, highlight the distinctive negative influences of parental rejection and family discord on these young people's mental health (Bouris et al. 2010; Bregman, Malik, Page, Makynen and Lindahl 2013; Ryan, Russell, Huebner, Diaz and Sanchez 2010). Experiences of family rejection can make these youth nearly six times more likely to report elevated levels of depression (Ryan, Huebner, Diaz and Sanchez 2009), and sexual minority youth in the general population have disproportionately high levels of mental health issues compared to heterosexual young people (Marshal et al. 2011). Young people who identify as gay and lesbian also exhibit higher levels of suicide ideation when they are closeted from their parents or their parents were unaccepting of their sexual orientation (D'Augelli 2002). Same-sex attraction can also reduce the benefits of positive parental relationships for sexual minority youth, as they continue to report heightened engagement with risky behavior such as drinking and drug use (Pearson and Wilkinson 2013).

In addition to mental health issues, gay and lesbian young people who experience residential instability may be at risk for further negative outcomes, as a link has been established between same-sex attraction and running away (Waller and Sanchez 2011). LGBT youth become homeless at disproportionate rates compared to heterosexual young people (Woronoff et al. 2006), with 20% of homeless youth identifying as LGBT compared to only 5-10% in the general population (Center for American Progress 2010). Relatedly, nearly 40% of young people who use street outreach services are LGBT (Durso and Gates 2012). Homeless youth who identify as lesbian and gay are at greater

risk for numerous issues compared to their heterosexual counterparts, including victimization and substance abuse (Cochran, Stewart, Ginzler and Cauce 2002; Tyler 2008). Sexual minority youth with a history of homelessness are also more likely to report mental health issues, such as depression, than heterosexual homeless youth (Whitbeck, Chen, Hoyt, Tyler and Johnson 2004) and their non-homeless LGB counterparts (Rosario, Schrimshaw and Hunter 2012). The stigma attached to LGBT identities can transcend numerous social contexts, including both those that are marginalized, such as homelessness, and those that are considered more normative, such as college campuses.

Experiences of discrimination and prejudice based on sexual orientation can be especially salient in the course of higher education as young people develop their identities and chart their future pathways (Perry 1999). Additionally, prejudiced attitudes regarding LGBT identities remain prevalent on college campuses, where heteronormativity, or the privileging of heterosexuality, dominates normative expectations (Wickens and Sandlin 2010). Sexual minority college students tend to endure more instances of harassment related to their sexual orientation from their peers (D'Augelli 1992; Woodford, Kulick, Sinco and Hong 2014), with approximately 36% of LGB college students reporting some form of on-campus harassment (Rankin 2003; Woodford, Kulick, Sinco and Hong 2014). Despite these challenges, young adults who identify as LGBT can find the college environment to be a crucial time of identity development and self-actualization regarding their sexual orientation (Bilodeau and Renn 2005; Stevens 2004). Therefore, the combined search for identity and experiences of harassment and discrimination has the potential to place LGBT college students at higher

risk for a number of mental health issues, such as depression and suicide ideation, compared to their non-LGBT counterparts (D'Augelli 1993; Westefeld, Maples, Buford and Taylor 2001).

While scholars have identified the unique experiences of LGBT college students (Sanlo 2004) and LGBT homeless young adults (Cochran et al. 2002) in separate research, there are few studies that explore how these two groups of young adults compare and contrast in domains such as identity development, coping strategies and family and peer relationships. While LGBT young adults in college and on the street may be similar in terms of chronological age, they typically navigate very disparate life course trajectories and pathways as a result of differential access to resources and opportunity structures (O'Rand 2006). For example, though the theory of emerging adulthood uniquely captures the time of exploration and self-actualization that ideally encompasses the period between adolescence and adulthood, it is very much a white, middle-class model of development that is often limited to the experiences of young people attending college (Arnett 2000, 2015). Thus, young people from marginalized social locations, such as homeless youth and LGBT youth, tend to be excluded from this conception of individual growth that emphasizes privileged backgrounds and social and cultural capital (Arnett, Kloep, Hendry and Tanner 2011). The present study seeks to fill these gaps in the literature by simultaneously exploring the distinctive life courses of both LGBTQ homeless and college young adults through their subjective interpretations of their experiences in the contexts of family, peers and identity formation. To do this, I qualitatively examine the following research questions: How do LGBTQ young adults navigate their current life course trajectories and family and peer relationships in various

social contexts? How do the experiences of homeless LGBTQ young adults and those enrolled in college compare and contrast along the lines of sexual and gender identity-related challenges?

In the following chapters, I situate my study within the realm of existing knowledge on LGBTQ young adults, outline my theoretical framework and describe the methodology. First, Chapter 2 synthesizes the extent of understanding surrounding LGBTQ homeless young adults and LGBT college students and the specific challenges they face in their social worlds. Next, Chapter 3 presents the life course perspective and demonstrates its utility in examining the social relationships and experiences of LGBTQ young adults. Then, Chapter 4 describes my data collection and recruitment strategies, as well as my plan to use narrative inquiry as an analytic tool. Chapter 5 presents the study findings to illustrate how LGBTQ young adults navigate their social contexts, identities and social relationships in unique ways. Subsequently, Chapter 6 outlines a discussion of the implications of this study's findings in relation to broader social issues, future research and service providers, as well as highlighting the study's limitations. Finally, Chapter 7 concludes by reiterating the importance of this study as it emphasizes the diversity of LGBTQ young adults and their life experiences.

# **CHAPTER 2**

# LITERATURE REVIEW

Runaway and Homeless LGBT Youth

Overall, young people with a history of same-sex attraction and/or relationships are considerably more likely to run away from home in relation to youth who report only opposite-sex experiences (Waller and Sanchez 2011). The majority of research on LGBT homeless youth utilizes a risk-based framework that emphasizes their on-the-street experiences while neglecting the influence of their family backgrounds and individual interpretations of their experiences. Primarily, the risks that lesbian, gay and bisexual (LGB) homeless youth face on the street are significantly exacerbated by their sexual orientation status (Cochran et al. 2002; Gattis 2013; Van Leeuwen et al. 2006). LGB-identified youth who are also homeless engage in more behavior that enhances their risk for contracting HIV, such as survival sex in exchange for food, shelter or other life essentials (Gangamma, Slesnick, Toviessi and Serovich 2008). Relatedly, gay and lesbian homeless youth are more likely to be treated for HIV (Rew, Whittaker, Taylor-Seehafer and Smith 2005) and experience higher levels of street victimization compared to their heterosexual counterparts (Cochran et al. 2002; Tyler 2008).

In addition to engaging in more at-risk behavior and undergoing more negative street experiences, LGBT homeless youth also experience more adverse mental health consequences (Cochran et al. 2002; Whitbeck et al. 2004). For example, LGB street youth were much more likely to exhibit symptoms for depression, post-traumatic stress syndrome and suicide ideation when compared to heterosexual homeless youth (Whitbeck et al. 2004). Concerning their life trajectories, LGB young people on the street

also have more pronounced histories of residing in a mental health facility (Noell and Ochs 2001), which could further contribute to their residential instability. Possibly as a means of coping with the harsh reality of street life and the added stigma of sexual orientation, LGBT homeless youth also report much higher levels of substance use and abuse (Cochran et al. 2002; Van Leeuwen et al. 2006). In a similar vein, LGBT young people with histories of homelessness also experience greater discrimination in homeless youth housing and other services based on their sexual orientation, which can further exacerbate the numerous other risk factors they face (Hunter 2008). Taken together, these unique issues faced by LGBT homeless youth, such as victimization and mental health problems, call for further research on their family relationships and how these social ties impact their health and well-being. The majority of research on LGBT homeless youth describes their experiences from an outsider's perspective and fails to account for the intricate role that sexual orientation plays in these young people's lives and how it shapes their social networks.

Runaway and Homeless Young People and Family Relationships

Familial conflict and discord are primary factors in shaping a young person's entrance into residential instability and potential homelessness (Kennedy et al. 2010; Thompson et al. 2010; Tyler 2006). Specifically, mixed, or blended families that include stepparents, extended family members and non-blood related kin have been shown to place youth at higher risk for homelessness by increasing household stress levels as family members adapt to each other (Mallett and Rosenthal 2009). Single-parent homes tend to experience higher rates of financial distress and limited parental monitoring, which can drive youth to homelessness (Finkelstein 2005). Young people living in

unstable, conflict-laden foster care and group home situations are also more likely to experience homelessness at some point in their lives compared to their counterparts living in a nuclear family, domiciled environment (Mallett et al. 2005; Thompson et al. 2010; Tyler and Melander 2010). Disruptions in family living situations such as divorce or death of a parental figure can also create pathways into homelessness for young adults (Kennedy et al. 2010).

Youth may also be pushed into running away and subsequent homelessness if they are repeated victims of physical, sexual or emotional abuse (Mallett and Rosenthal 2009; Tyler 2006; Tyler and Whitbeck 2004). One study found that 95% of homeless youth had suffered some form of physical abuse at the hands of a primary caregiver prior to leaving home (Tyler and Melander 2010). Specifically, approximately 69% of perpetrators who physically abused homeless youth can be categorized as a biological mother and/or father (Tyler and Cauce 2002). Youth are also at increased risk of running away at an earlier age if they experienced neglect or sexual abuse in their household (Ferguson 2009; Thrane et al. 2006). Experiences of personal maltreatment and victimization tend to be positively correlated with the number of instances a youth initiated running away (Tyler and Bersani 2008). These abusive experiences are often not isolated as singular experiences, but their negative effects reverberate throughout the young person's life.

Psychological issues such as depression and suicidal ideation may develop over time as a result of repeated familial abuse and maltreatment, thus diminishing a youth's capacity to cope with life's struggles (Ryan et al. 2000; Thompson et al. 2010).

Specifically, it has been shown that familial sexual abuse is highly correlated with

thoughts of suicide among homeless youth (Yoder, Hoyt and Whitbeck 1998). Consequently, physical and sexual abuse in early childhood can lead a youth to utilize violence-laden survival strategies once on the street, thereby replicating a form of socialized violence (Kennedy et al. 2010). Additionally, young women who experience early childhood sexual abuse are more at risk of developing depressive symptoms, which can in turn make them more vulnerable to future victimization within the street context (Chen, Tyler, Whitbeck and Hoyt 2004). Witnessing domestic violence can also damage a youth's sense of efficacy even if he or she is not the direct target of violence, thereby solidifying the path to homelessness if a young person feels they must flee an abusive home (Tyler 2006). Given the increased risks that LGBT homeless youth face and the conflicted family backgrounds of homeless youth in general, more research is needed to understand the specific experiences of LGBT homeless youth and their multifaceted family dynamics.

# LGBT College Students

Similar to, yet distinctive from, LGBT homeless young adults, sexual minority college students also face risks and challenges related to their sexual orientation in the unique social context of campus life (Rankin 2005). For example, approximately 74% of LGB students attending college labeled their campus climate as homophobic, while 60% of these young people opted to remain closeted so as to reduce their chances of peer and structural discrimination (Rankin 2003). More recently, Woodford and colleagues (2014) found that 36% of lesbian, gay, bisexual or queer (LGBQ) college students had experienced some form of sexual orientation-related victimization, such as verbal or physical assaults. While they can experience a wide variety of on-campus harassment,

LGBQ students are 17 times more likely to endure verbal derogation than any type of physical attack (Rankin, Weber, Blumenfeld and Frazer 2010). Encounters with campus homophobia can distinctively impact sexual minority college students, as they are significantly more likely to report both anxiety and depressive symptoms compared to heterosexual students (Woodford, Han, Craig, Lim and Matney 2014).

LGBT college students may also use substances as a means of fitting in, as one study found that gay men were much more likely to report drinking alcohol to avoid social exclusion when compared to heterosexual students (Longerbeam, Kurotsuchi Inkelas, Johnson and Lee 2007). Compared to their heterosexual counterparts, LGB college students report higher levels of drug and alcohol use, as well as more negative consequences related to alcohol use (Kerr, Ding and Chaya 2014).

In light of these unique barriers, LGBT young people also experience distinctive identity transformations during their college years (Stevens 2004). Processes of identity formation take on many unique forms and intersect with many social domains, making the concept of identity especially complex in the college environment as students simultaneously inhabit numerous roles and statuses (Abes and Jones 2004; Abes, Jones and McEwen 2007; D'Augelli 1991). Sexual minority college students may also express more fluid identities if they do not conform to heteronormative expectations of roles and statuses in college, resulting in a challenge to prevalent norms regarding gender and sexuality (Abes and Kasch 2007). For example, young adult undergraduates who increase their involvement in LGBT-related groups and activism on campus can experience an increasingly public sexual minority identity that develops over the course of their college experience (Renn 2007). Lesbian and gay college students also struggle to reconcile their

multiple identities with their sexual orientation, such that the college context prompts them to engage in processes of self-actualization that can result in experiences of identity conflict (Love, Bock, Jannarone and Richardson 2005).

LGBT Young Adults and Family Relationships

Despite the struggles of college campus life and sexual orientation identity formation, peer and family social networks can serve as critical sources of support for all LGBT young adults in maintaining positive mental and emotional health (Goldfried and Goldfried 2001; Nesmith, Burton and Cosgrove 1999). Research has primarily highlighted the descriptive characteristics of LGBT young adults' family relationships, with little attention given to the intricate dynamics of how youth navigate familial and peer networks. For example, perceptions of general social support, including that from friends and family, can lower LGBT college students' levels of depression at the same time that it can improve their overall life satisfaction (Sheets and Mohr 2009). Increased feelings of family acceptance among LGBT young adults can also act as a protective factor against adverse mental health outcomes such as depression and suicide ideation (Ryan et al. 2010). Conversely, parental awareness of a young person's non-heterosexual orientation may increase their experiences of sexual orientation-related verbal abuse from their parents (D'Augelli, Grossman and Starks 2005), which may be why perceptions of friends' support are more predictive of a youth's disclosure of their sexual orientation (Shilo and Savaya 2011). LGB young adults are also more prone to suicidal attempts if their parents have actively discouraged them from displaying their gender nonconformity (D'Augelli et al. 2005). Based on the complex family dynamics experienced by LGBT young adults, further research is needed to unpack how young people interpret and

manage their family relationships in the context of their sexual identities and social environments.

LGBT Identity as a Link Across Social Contexts

Experiences of social harassment and discrimination have the potential to transcend differing social contexts in their critical influence on young people's mental health. For example, LGBT individuals often experience microaggressions, or interactional exchanges that derogate one's non-conforming gender or sexual identity, and these could occur in a variety of social settings. Sexual minority young adults in the general population (Nadal et al. 2011) and as students on campus (Woodford et al. 2014) specifically endure high levels of heterosexist microaggressions that adversely impact their mental health. The socially ubiquitous phrase "that's so gay" has been linked to adverse social and physical health outcomes for LGB college students, ranging from feelings of isolation to chronic headaches (Woodford, Howell, Silverschanz and Yu 2012). Transgender individuals face unique social stigmas that can range from transphobic discourse to disapproval and family rejection (Nadal, Skolnik and Wong 2012). Similarly, homeless LGBT young adults face discrimination related to their gender and sexual identities in shelters, which can inhibit their sense of safety in these facilities (Hunter 2008). In these ways, young people's LGBT identities can act as a link across unique social contexts.

Microaggressions and homophobic bias can also create a hostile academic climate for LGBT young people, especially when teachers and other school officials are complicit in these behaviors (McCabe and Dragowski 2013). Academic relationships are critical for marginalized youth, as lesbian, gay and bisexual adolescents without a

connection to a school official exhibit elevated levels of mental health troubles, such as suicide ideation (Seil, Desai and Smith 2014). Sexual and gender minority young people also endure distinct experiences of school victimization based on their sexual orientation, such as bullying and exclusion (Russell, Ryan, Toomey, Diaz and Sanchez 2011). Specifically, LGBT homeless youth face greater difficulties in completing high school when they are disconnected from academic support systems (i.e. counselors) and their schools lack LGBT-friendly programs (Bidell 2014). School victimization can severely impede LGBT young adults' psychosocial functioning as well as their potential academic achievement (Toomey, Ryan, Diaz, Card and Russell 2010). For these young people, school-related support, such as Gay-Straight Alliances, could be indicative of wider community resources (Hackimer and Proctor 2015) and social contexts that could be integral to the well-being of LGBT students as well as homeless LGBT young adults.

#### **CHAPTER 3**

# THEORETICAL FRAMEWORK

As introduced by Elder (1998), life course theory is a valuable framework for assessing the situations and circumstances that shape young people's life trajectories, such as experiencing homelessness or attending college, and the effects of these turning points continue to influence young people's outcomes throughout their lives. This theory is understood through its recognition that "early transitions can have enduring consequences by affecting subsequent transitions, even after many years and decades have passed" (Elder 1998:7). In this way, an individual's life events create a chain of opportunities, disadvantages or some combination of the two possibilities, which result in multiplicative effects. For example, a young adult may reap further benefits of attending an elite school such as guaranteed employment after graduation (Howard and Gaztambide-Fernandez 2010), while conversely, youths may continue to fall behind if they prematurely drop out of high school and are confined to low-wage work, or experience bouts of homelessness (Whitbeck and Hoyt 1999). As such, positive or negative life events can either improve or exacerbate present conditions as well as future opportunities for young adults.

The tension between structural limitations and an individual's sense of personal control over their lives is also integral to a life course perspective of LGBTQ young adults' experiences. A young person's sense of agency, or the perceived ability to enact purposeful decisions to a degree within constraining structures, is closely tied to his or her level of identity development, such that young people with a more established sense of identity exhibit elevated levels of agency (Schwartz, Côté and Arnett 2005). Similarly,

agentic young adults feel responsible for controlling their own life course trajectory and believe that they have control over their decisions and outcomes by possessing the ability to overcome challenges on their life course pathway (Côté and Levine 2002). For sexual minority youth in particular, perceiving social support and connectedness to their peers and the broader community is significantly related to an enhanced sense of agency (Poteat, Calzo and Yoshikawa 2016). Therefore, understanding the varying life course pathways of LGBTQ young adults is key in exploring the unique ways that they make choices in a cultural climate that both constrains and resists non-normative gender and sexual identities.

Life course theory is useful for exploring individual transitions and trajectories that shape future outcomes through a lens that simultaneously considers the influence of social, historical and cultural contexts (Elder 1998). The social dynamic of "linked lives" of individuals helps to explain how family members and peer groups inhabit interconnected trajectories that bidirectionally influence one another (Elder 1994; Moen and Hernandez 2009). The quality of family relationships can have lifelong impacts, as disruptions and conflict in early family life between parents and their children continue to exert influence over this relationship well into the children's adulthood (Whitbeck, Hoyt and Huck 1994). A family unit in discord can drastically alter a youth's life, leading to detrimental consequences for his or her social and emotional development (Cavanaugh and Huston 2008).

Relatedly, processes of identity formation are not static and often shift in nature and purpose throughout life course trajectories (Caspi and Roberts 2001). Life course experiences also greatly hinge on individual social locations and the ways that these

statuses can influence outcomes and opportunities (O'Rand 2006), such as stigma and prejudice directed at sexual minorities and how this can limit life prospects (Meyer 2003). Family relationships are also integral to understanding life course processes among LGBT individuals as sexual orientation coming out trajectories can lead to complex family dynamics (LaSala 2010) and these negotiations can continue throughout the life course (Denes and Afifi 2014). For example, young people coming out as gay and lesbian to their family networks often strategically manage this process by carefully planning the timing, context and language used with family members (Newman and Muzzonigro 1993; Valentine, Skelton and Butler 2003). Family reactions to this process and levels of social support can also impact familial relationships through experiences of rejection or affirmation of the LGBT individual's identity (LaSala 2010).

# **Life Course Theory and Homeless Young Adults**

Life course theory has been used in previous research to identify pivotal developmental stages in an adolescent's life, and how adverse social, economic and environmental forces may ultimately result in a youth's homelessness (Kennedy et al. 2010; Tyler 2006). Life course events occurring out of order based on socially normative expectations, such as giving birth to a child before graduating high school, can result in detrimental outcomes for the individual (Elder 1998; Hoffman, Foster and Furstenberg 1993). The damaging effects of these "off-time" life course transitions may be most salient for young people if they occur in formative developmental periods (Elder, Johnson and Crosnoe 2003). For example, youths' family backgrounds and specifically timed "events," such as instances of parental drug use or child abuse, can lead to unstable child developmental trajectories and a youth's exit from a housed environment through

formal institutional removal or the young person's running away (Tyler 2006).

Additionally, early exposure to familial violence and caregiver rejection can act as pivotal experiences paving the way towards young people's premature independence and homelessness (Kennedy et al. 2010). Research agrees that pathways leading to homelessness often begin in an individual's formative years of childhood and adolescence, thus setting the stage for adult life trajectories (Lee, Tyler and Wright 2010; Tyler 2006).

# **Life Course Theory and College Students**

The experience of college, especially for traditional students between the ages of approximately 19 and 25, can be a period of self-discovery and identity formation that presents unique opportunities for life course turning points, or changes in trajectories that have lifelong impacts (Elder et al. 2003; Tanner 2006). For example, emerging adulthood has been conceptualized as a process by which young people, primarily of traditional college student age, engage in processes of self-actualization as they reorient their life trajectories during a time of uncertainty to establish a sense of independence (Arnett 2000). Specifically for LGBT college students, the coming out process can act as a critical life course milestone that shapes their identity, social networks and developmental trajectories (Evans and Broido 1999; Rhoads 1994). Younger generations are also becoming more resistant to sexual and gender identity labels, and often opt out of claiming a particular identity and prefer to view sexuality and gender as more fluid components of themselves that defy categorization (Savin-Williams 2011).

Examining youths' family backgrounds and identity development through a life course perspective is also an effective framework for comparing trajectories of young

adults who share certain characteristics and yet are pursuing qualitatively different developmental pathways. Specifically, this study aims to examine the life course trajectories of two groups of young people who are similar in terms of chronological age, but inhabit disparate routes of development: homeless youth and college students. This research will explore the complex diversity of LGBT young adults by comparing and contrasting the experiences of these young people from their own perspectives, while also highlighting the importance of their linked lives.

# **CHAPTER 4**

# **METHODOLOGY**

Sampling Procedure

Participants for both samples self-identified as lesbian, gay, bisexual, transgender, queer or another non-normative gender and sexual identity and were between the ages of 19-26. This narrow age range was chosen based on the critical life course transition from adolescence to young adulthood, often termed emerging adulthood that represents a pivotal turning point in young people's lives (Arnett 2000). All data were collected in the Midwest. Though lesbian, gay and bisexual (LGB) refer to sexual identities, transgender (T) denotes a gender identity and queer (Q) can refer to both, I include these identities together based on their societal subjugation in reference to heterosexuality and gender-conformity (Jackson 2006). To promote inclusivity, additional sexual and gender minority identities were welcome to participate, such as asexual and pansexual, however the focus remained on LGBTQ young adults in accordance with the majority of previous research focusing on individuals with these identities (Ryan et al. 2010).

Distinct procedures were utilized to gain access to each population of interest. While there is potential overlap between these two populations in the form of college students experiencing homelessness, I screened out these individuals to maintain the focus of this study in examining distinctive social contexts. I recruited participants for the college student sample by advertising the study throughout campus bulletin boards and email listservs such as those distributed for the Women's and Gender Studies program as well as the LGBTQA+ Resource Center and Women's Center. To participate, young adults in the college sample had to be enrolled in an institution of higher education.

Homeless young people were recruited using a non-probability method of convenience and snowball sampling once key informants were identified, such as members of particular friendship groups (Creswell 2013; Merriam and Tisdell 2016). The definition of homelessness requires that the youth currently reside in a shelter, on the street, or independently because they had run away, had been pushed out, or had drifted out of their family of origin (National Center on Family Homelessness 2011). This portion of the sample was recruited from agency shelters and transitional living programs that provide services to homeless youth and young adults.

# Researcher Positionality

Furthermore, the role of the researcher must also be considered within the contexts of the college and homeless environments and sexual minority identities. The sensitive subject matter of the interviews (i.e. sexual orientation) required careful consideration of how questions were asked and subsequently interpreted (Lee and Lee 2012). As a gender-conforming female graduate student whose sexual orientation was unknown to participants, I remained cognizant of how my positions of social privilege influenced the interviewing dynamics of LGBTQ-identified individuals (Levy 2013). To gain access to gatekeepers and key informants within these hidden populations, I strived to develop rapport with participants and allowed them to be active participants in the research process (Couch, Durant and Hill 2012) by developing my own sense of cultural competence in working with LGBTQ communities (Van Den Bergh and Crisp 2004). Boundary issues arose, however, when some participants inquired as to my personal life or requested favors such as car rides or money for transportation (the latter primarily from homeless young adults). This challenge prompted me to enact strategies that

maintained the relationship of researcher-participant when these boundaries became blurred, such as deflecting personal questions about my sexuality to pointed questions about the participant's own identity and providing respondents with service agency referrals when they asked for my advice on identity-related issues (Dickson-Swift, James, Kippen, Liamputtong 2006). Throughout the recruitment and data collection processes, I reflexively challenged my own assumptions about the marginalized groups of both LGBTQ young adults and homeless youth and my own status as the researcher so as to enhance the authenticity of the young people's stories (Reed, Miller, Nnawulezi and Valenti 2012).

# Data Collection

I conducted all of the interviews. All study participants completed one digitally tape-recorded, in-depth face-to-face interview lasting approximately one hour and a short demographic questionnaire. LGBTQ college students were interviewed at a private location that was convenient and comfortable for them, such as a reserved room at a public library or a small conference room. Homeless young adults were interviewed in agency shelters and transitional living programs that provide services to homeless youth and young adults, as well as private study rooms at public libraries. Study procedures were explained to all participants and informed consent was obtained prior to the start of the interview, whereby both the participant and researcher provided their signatures. All participants received \$20 compensation upon completion of the interview in exchange for their time.

Conceptualization and Measurement

Within the format of the semi-structured, face-to-face interviews, respondents were asked a series of open-ended questions that revolved around the topics of LGBTQ identity and family relationships. The same questions were used for all participants, so as to compare and contrast the experiences of LGBTQ college and homeless young adults. The following research questions guided the study:

- 1. How do LGBTQ young adults navigate their life course trajectories and family and peer relationships in various social contexts?
- 2. How do the experiences of homeless LGBTQ young adults and those enrolled in college compare and contrast along the lines of sexual and gender identity challenges? Additionally, respondents were asked to fill out a short demographic form before the start of the interview. These research questions elucidate the unique experiences of LGBTQ young adults and how they compare and contrast across the disparate social contexts of the college campus and homelessness-related environments.

# Ethical Concerns

The institutional review board approved this study. Prior to all interviews, I carefully reviewed an informed consent form with all respondents and they all provided their signatures (or initials to preserve confidentiality) if they consented to the study requirements. To ensure confidentiality, all respondents were assigned pseudonyms in the interview transcriptions. These pseudonyms are used in the present study to ensure continued respondent confidentiality. Furthermore, all participants were told that they can stop the interview at any time or skip over questions they did not feel comfortable answering. LGBTQ college students were provided with resources available on campus

(i.e. counseling services), while homeless young adults were given a list of services and agencies available to homeless youth (i.e. street outreach, transitional living).

Data Analysis

I performed all data analyses using the computer software MAXQDA, a qualitative data analysis program. To begin, all interviews were transcribed verbatim (including pauses and filler words such as "uh" and "um") to ensure that the true nature and meaning of the participants' words were captured (Oliver, Serovich and Mason 2005) and Word documents of these transcriptions were uploaded into MAXQDA. Drawing from Charmaz's (2014) constructivist grounded theory coding procedures consistent with my worldview and research design, I first utilized the method of initial coding to determine emergent themes and categories that corresponded with concepts of interest, such as LGBTQ identity formation and family relationships (Charmaz 2014). Next, I employed focused coding to hone in on the participants' subjective interpretations of their life course experiences. The final themes emerged inductively from the data. The combination of initial and focused coding allows for a constructivist perspective to emphasize the participants' understandings of their lived realities and the meanings they attach to their experiences (Charmaz and Belgrave 2002).

Validity was assessed by building evidence for a code or theme (e.g., LGBTQ identity formation, dynamics of family and peer relationships) from several respondents and through the collection of demographic information as well as the rich data from the in-depth interviews (Berg and Lune 2004; Creswell 2013). To examine intercoder reliability, a predetermined coding scheme and qualitative codebook was used to identify whether or not the same codes were assigned between text passages for both the LGBTQ

college student and homeless young adult transcripts (Boyatzis 1998). I also documented the chain of formulated interpretations throughout the data analysis process (Angen 2000) using an audit trail to illustrate how codes and overall themes were constructed (Padgett 2008). To enhance the credibility of the findings, participants had the option of being recontacted so that they may engage in the process of member checking, whereby I asked them to review the raw data and assess the accuracy of assigned codes and themes (Creswell and Miller 2000). Of all the participants, 18 college students (75%) and 10 homeless young adults (45%) provided feedback over email on both the accuracy of their interview transcripts and earlier versions of the themes that included brief, generalized descriptions of the major codes and subcodes. Member checking improved the validity of the raw data as well as the findings by allowing participants to ensure their intended meaning was captured and that my interpretations were legitimate. Finally, a collaborative data conference was held with other experienced qualitative researchers to assess the overall validity of the findings. All of these strategies enhance the rigor and trustworthiness of this study (Creswell 2013; Padgett 2008).

# Narrative Inquiry

Narrative inquiry was used to examine the experiences of LGBTQ young adults on their pathways to identity formation as well as the dynamics of their social relationships. A narrative analysis approach is ideal for exploring how individual lives are shaped by their social locations and social networks, as well as how these experiences can shift across the life course (Creswell 2013; Elder et al. 2003). Specifically, this analytical method allows for an examination of young adults' life course trajectories "as expressed by the unfolding of a specific sequence of events" (Franzosi 1998:520). In

terms of LGBTQ young adults, narrative inquiry helped illuminate the unique experiences of sexual minorities as they recount their processes of identity formation in the context of their social lives through a connected, chronological framework (Riessman 2008). Furthermore, narrative inquiry takes into consideration the distinctive contexts of individual stories and the ways that environments shape people's experiences (Loh 2013), which is useful for a comparison of the campus and street contexts in LGBTQ young adults' lives. This method focuses directly on the young adults' individual perspectives to demonstrate the vitality of the findings as a whole (Arnett 2005).

# Respondent Characteristics

My sample included 24 women (52%), 19 men (41%) and three (7%) respondents identified outside of the gender binary, such as bigender or genderqueer. Of respondents who identified as either a woman or a man, eight participants identified as transgender. As participants were allowed to self-identify however they chose, those who identified as transgender also identified as either a woman or man, so these gender categories are not mutually exclusive. Concerning sexual orientation, seven people identified as lesbian (15%), 11 were gay (24%), 20 were bisexual (43%), four were queer (9%) and four people identified outside of the LGBQ spectrum, such as asexual. One half of respondents identified with their sexual orientation for 1-4 years, 10 young people (22%) for 5-8 years and 13 people (28%) reported 9+ years. Ages ranged from 19 to 26 years. Thirty-two respondents were White (70%), five African American (11%), two Asian-American (4%) and seven bi-racial or multi-racial (15%). Regarding relationship status, 21 participants were single (46%) and 20 were in a relationship (43%). Table 1 provides participant demographics by their primary social context (i.e., college or homeless).

**Table 1. Respondent Demographics by Primary Social Context** 

	College	Homeless	Total
	(N=24)	(N=22)	(N=46)
Age	(11-24)	(11-22)	(11-40)
Mean	21	21	21
Range	19-26	19-26	19-26
Race/ethnicity	19-20	19-20	19-20
White	21 (88%)	11 (500/)	22 (700/)
African-American	0	11 (50%)	32 (70%) 5 (11%)
		5 (23%)	
Asian-American	1 (4%)	1 (4%)	2 (4%)
Bi/Multi-racial	2 (8%)	5 (23%)	7 (15%)
Sexual orientation	2 (120/)	4 (100/)	7 (150/)
Lesbian	3 (12%)	4 (18%)	7 (15%)
Gay	5 (21%)	6 (27%)	11 (24%)
Bisexual	9 (37%)	11 (50%)	20 (43%)
Queer	4 (17%)	0	4 (9%)
Other	3 (13%)	1 (5%)	4 (9%)
How long identified as that sexual orientation	16 (670/)	7 (220/)	22 (500()
1-4 years	16 (67%)	7 (32%)	23 (50%)
5-8 years	3 (12%)	7 (32%)	10 (22%)
9+ years	5 (21%)	8 (36%)	13 (28%)
Gender*			
Woman	13 (54%)	11 (50%)	24 (52%)
Man	8 (33%)	11 (50%)	19 (41%)
Transgender	4 (17%)	4 (18%)	8 (17%)**
Other	3 (13%)	0	3 (7%)
Relationship status			
Single	11 (46%)	10 (45%)	21 (46%)
In a relationship	10 (42%)	10 (45%)	20 (43%)
Married	0	2 (9%)	2 (4%)
In a domestic partnership	2 (8%)	0	2 (4%)
Other	1 (4%)	0	1 (2%)
Region			
Urban	17 (71%)	17 (77%)	34 (74%)
Suburban	7 (29%)	5 (23%)	12 (26%)
Family class status growing up			
Working class	7 (29%)	11 (50%)	18 (39%)
Middle class	13 (54%)	10 (45%)	23 (50%)
Upper-middle class	4 (17%)	1 (5%)	5 (11%)
Upper class	0	0	0
Parental education			
Some high school	1 (4%)	4 (18%)	5 (11%)
Graduated high school	4 (17%)	7 (32%)	11 (24%)
Some college	3 (13%)	3 (14%)	6 (13%)
Associate's degree	1 (4%)	1 (4%)	2 (4%)
Bachelor's degree	5 (21%)	4 (18%)	9 (20%)
Professional or advanced degree	10 (42%)	1 (5%)	11 (24%)
Other (GED)	0	2 (9%)	2 (4%)
Public assistance growing up			
Yes	6 (25%)	15 (68%)	21 (46%)
No	18 (75%)	7 (32%)	25 (54%)
Public housing growing up	` ′		
Yes	0	7 (32%)	7 (15%)
No	24 (100%)	15 (68%)	39 (85%)
10 11 1	(	- (/*)	( /-/

<sup>\*</sup>Categories not mutually exclusive \*\*Out of participants who identified as either a woman or a man

### **CHAPTER 5**

#### **FINDINGS**

For both LGBTQ homeless young adults and college students, sexual and gender identities emerged as complex phenomena that required young people's careful consideration and negotiation. In many ways, young adults' primary social context, which included either being in college or being homeless, shaped the ways that they could develop and conceptualize their LGBTQ identity by facilitating or obstructing this process, resulting in the first theme of "Identity as Privilege." Similarly, young people were steadfast in how they revealed or concealed their LGBTQ identity by carefully considering how it would be perceived, which encompasses the second them of "Strategic Identity Management." Finally, while all of the LGBTQ young people, regardless of social context, cited challenges and difficulties they had faced in their lives, they were also discerning in their experiences, whereby they reframed negative experiences into positive ones, embodying the third theme of "Identity as Life" Enhancement." These three themes underscore the complex ways that young adults embrace their gender and sexuality, and how their experiences are influenced by social context, relationships and personal strength.

# **IDENTITY AS PRIVILEGE**

The importance of identity and the influential nature of social context emerged as a central theme among both LGBTQ college students and LGBTQ homeless young adults. In developing their sexual and gender identities, young adults navigated their social contexts as a means of seeking out resources, support and opportunities that help them develop their self-concepts and coping strategies. For LGBTQ young adults,

however, particular social environments created unique experiences that shape their lives in varying ways depending on if they are integrated into a college campus context or if they are currently homeless and interacting with different service agencies. The following section highlights how social contexts shaped young adults' LGBTQ identities and the ways in which environment acts as a mechanism that largely privileges and supports identity development or hinders and creates obstacles to identity formation. For example, the majority of college students' descriptions of their educational context can be described as the following subtheme: "Campus Contexts as Conducive to Identity Development." While homeless young adults also recount complex encounters with identity formation, their social environment imposes complex barriers to introspection, which can be summarized as the subtheme "Homelessness as a Hindrance to Identity Development."

# **Campus Contexts as Conducive to Identity Development**

For LGBTQ college students, the campus context often represents a socially liberal atmosphere that is conducive to identity development across several different domains. Having access to LGBTQ-related resources and support networks is critical for young people in helping them develop their identities. While for many students the college experience is one that broadens their view of the world and exposes them to supportive social relationships, LGBTQ young adults' must also navigate the intricacies of the campus context when support and acceptance of their identities is not as readily available.

Social Support and Access to Resources

In addition to pursuing degrees in higher education, LGBTQ college students also use their time in college to develop their multiple intersecting identities, including gender and sexuality. For the majority of college students in this study, social support and campus resources specifically geared towards LGBTQ students and issues characterize their college experience. One such example is Charlie (pansexual demi-girl), who believes that her major in the college of Arts and Sciences has helped her to feel accepted in the classroom environment when discussing controversial issues:

I feel a sense of protection...I mean we're talking about the issues themselves and what the viewpoints are on either side and so I feel like it's a byproduct of the degree and the way it's set up by teachers for creating a safe space because I've had professors who I felt more comfortable around and I think that if anybody were to actually say anything, they would say something back and it totally wouldn't be tolerated.

Similarly, Gabriel (queer transgender man) shares a positive view of the college campus context in supporting his non-normative gender identity:

I really lucked out going to a liberal arts college and everybody is just very open minded and then I majored in [Arts & Sciences]...it is probably the best department to have support and people around you who understand that kind of thing and I just sent out an email to all my professors. A few of them who I felt really close to I talked to them in person and they were very supportive in helping me find any resources I need on campus and whatever to get everything the way it needed to be.

By finding support from and feeling accepted by their professors, these college students recounted positive experiences that helped them access resources and create a sense of belonging within the campus environment.

In addition to perceiving institutional acceptance of their LGBTQ identities, college students also emphasized the importance of peer support from classmates and friends they met on campus. Rylan (pansexual bigender) recalled how their transition into college allowed them to be open with their identity:

It's definitely a lot more open here [at college]. I'll talk to other kids and see what they are going through and we'll bounce things off one another and a lot of my friends are 'yeah, I think that's great you're being who you are.' I'm like 'okay, if these people accept me, why not just be me.' It has definitely had a hugely, hugely positive impact being here, being a college student.

Being surrounded by similar others and establishing relationships with other young adults who share their experiences is also critical for college students as it builds a sense of solidarity that can be key in helping young people feel accepted in social environments. For example, Samantha (bisexual woman) describes her perception of peer support on campus in the following way:

But now that I am in college, it is completely different here because I feel like we have a really diverse campus and there are a lot of people like me here. I feel like I've met a ton of friends, like [friend's name] and all of her friends I hang out with.

Creating supportive social networks is especially important for LGBTQ college students as they navigate wider societal conceptions that marginalize non-conforming gender and sexual identities

Broadly speaking, many LGBTQ college students perceived the college campus environment to be protective and inclusive of diversity so that they felt comfortable to express their gender and sexual identities without fear of blatant prejudice and discrimination. One such example is Natalie (bisexual woman), who stated, "I think that for the most part it is pretty good. It's better on campus than it is off campus for sure just because of the university setting. Like education tends to make you less homophobic, go figure." In another example, Phillip (gay man) emphasizes academic structures that protect LGBTQ identities and how this contrasts with the lack of legal protection he faces in the workplace in this particular region: "I think that being a student is probably better than if I was in the work force. I mean there's less challenges…well I mean the

university has protective codes, I mean. I feel safer here than I would elsewhere..."

Constructing the university campus setting as protective shows how LGBTQ college students can perceive wider institutional support from their college and utilize campus as a safe haven within the larger community.

Notably, several students specifically contrasted their experiences on campus to those they had in middle and high school, in that they struggled with social acceptance throughout their intermediate and secondary education years. Lucy (queer woman) described her transition from high school to college as transformative in allowing her to more fully embrace her LGBTQ identity:

We have a pretty accepting population and we've got groups and benefits and things like that so I think that it's nice to be a student at [university]. You're able to go to these places and feel okay where back in the day like in high school and stuff you didn't have those things, especially my high school being out in the country. I tried to get a gay/straight alliance group going and they [high school] denied that...so I think being in college it's a lot different because you experience an array of different people and different backgrounds...I think it's given me a different view and perspective about life....it's helped more than hindered me as a student.

Paul (gay man) shares a similar sentiment in the positive way he believes his transition to college shaped his life:

High school to college... oh man, that got even better. It just keeps getting better because at that point you know there's so many people on [university] campus who I can be friends with and I can be friends with guys and not have crushes on them but then there's a lot more gay people at [university] campus and so, you know, there's a lot better chance that if I start to develop feelings for someone that they might even reciprocate those...

Both Lucy and Paul's experiences illustrate how the dynamic life course transition into college can open up opportunities for LGBTQ college students in forming supportive peer and romantic relationships.

Learning about Identity in College

For many LGBTQ college students, the college experience also represented a crucial period in their lives when they were able to learn about their gender and sexual identities in an environment that encouraged critical thinking. Considering both the positive and negative aspects of analyzing her LGBTQ identity, Clark (queer woman) discussed the challenges she faced as she encountered more complex ideas surrounding identity politics:

And now, as I've been going through college I think I've had a lot more trouble because as I learn more, I sort of have to pause more and be like 'it's not quite that easy' or 'it isn't as simple as people make it out to be' or 'it's not just born this way.' There are things that we need to talk about when we talk about sexuality that we don't a lot of the time. Like class and race and gender.

For Clark, then, learning about her identity in college helped to broaden her view of the complexity of sexuality by encouraging her to incorporate more intersectional perspectives. Engaging in critical discourse in a supportive context, such as the college classroom, can help LGBTQ young people question their preconceived notions surrounding gender, sexuality and other socially marginalized statuses.

Additionally, experiences in college can help students establish a sense of solidarity in knowing that they are not alone. Alex (gay transgender man) summarized his moment of clarity during a class period in a diversity course:

About a year ago, a little bit more, I started taking Women's and Gender Studies classes and learning about what stuff meant, you know. So we learned about trans people and I'd be like 'oh that seems kind of familiar to my life experiences.' And that was pretty cool, just knowing that I wasn't so weird after all.

In addition to experiencing feelings of enlightenment in the classroom, other LGBTQ college students specifically sought out classes that related to their gender and/or sexual identity. For example, Rylan described how being more open about his identity helped him to feel more comfortable presenting in non-gender conforming ways (i.e. "I'm going

to get my nails done for the first day of class") and pursue gender and sexuality studies: "I picked up an LGBT minor that we're now offering here and one of my classes is by a professor that I really looked up to. He's a trans male...so I'm going to be a lot more open." By actively seeking out opportunities to learn more about themselves and broader social issues, LGBTQ college students like Rylan can enhance their own feelings of self-acceptance and social belonging. In these ways, the campus context creates a setting that is conducive for LGBTQ college students to explore the meaning of their individual identity and in so doing, learn more about issues related to social diversity.

# LGBTQ Activism and Advocacy

Opportunities for LGBTQ activism and advocacy on campus were also key in developing college students' sense of their identities and their role in the larger LGBTQ community. Many students became involved in campus groups and activities that helped them engage in efforts aimed at improving the lives of LGBTQ individuals. Alex developed a passion for educating others about LGBTQ issues after participating in an LGBTQ speakers' panel:

I'll go to a panel and there will be a gay guy and maybe a lesbian girl and then me. Then I get all the questions because they'll be like 'oh my god, you're trans?' But I really, really like doing that because it gives me a chance to educate people that have no idea what being trans is about. So that's really, really rewarding for me...educating people that want to know and want to make campus a more inclusive environment for you, that's really awesome.

Another student, Steven (gay man) described his experiences on the LGBTQ panel as one that helps to push people beyond their comfort zones and challenge their beliefs:

It's really cool community education and I really enjoy both getting to enhance the knowledge and insight of other people about the issues that face the LGBTQQA community and then also seeing some of the uncomfortability is a little enjoyable as well. Like people don't know how to approach these kinds of things and they don't know what questions are too far. Generally when I'm on a panel, no question is too far. I don't really care what you ask me.

By engaging with opportunities for LGBTQ activism and advocacy on campus, college students work to develop their own identities at the same time that they are afforded the chance to engage in the fulfilling experience of peer and community education.

In addition to practicing LGBTQ activism and advocacy through campus opportunities, several college students also engaged with activities in the wider community as a way to expand their role as advocates. Gabriel, for example, strongly believed in helping other LGBTQ individuals, particularly trans people, and was eager to share his experiences if it had the potential to improve someone's life:

For me I don't think it's possible for me to ever not be open about being trans because I want to be there to help others. I know a lot of people that, you know, they just want to transition and be done. Like not ever bring it up again. But for me it's very important that I am there to kind of help others navigate what they're going through and kind of be a role model.

Despite the challenges that LGBTQ young adults face, college students like Gabriel reframed their experiences as having an important purpose, such as being "a role model" for other trans people who are struggling. Some students, like Steven and Alex, highlighted the importance of LGBTQ activism on the college campus, while others, like Gabriel, showed how extending that sense of advocacy to the broader community is personally fulfilling. Taken together, forms of LGBTQ activism on the college campus and the larger community is oftentimes a key source of meaning for LGBTQ college students.

Managing Prejudice and Discrimination on Campus

Some students involved in subcultures of the wider campus community noted that while the college environment as a whole is accepting of LGBTQ individuals, they remained cognizant of how particular campus groups may respond to non-conforming gender and sexual identities. Jake (bisexual man) is especially aware of the heteronormative nature of Greek life and he takes this into consideration when interacting with his peers:

In the Greek community you can tell it is pretty toxic. In [Midwestern state] from what I can tell it is pretty homogeneously heterosexual and what have you. I'm not out to any of my Greek friends, at least people I know are Greek, and I don't know if I would be necessarily comfortable with it, at least not right now. Maybe that will change but until then I can kind of sense toxicity with that part of campus.

Furthermore, some students distinguish across different majors and academic departments in terms of levels of acceptance and comfort with disclosing their LGBTQ identity. Bethany (bisexual woman) recalled how being around students with particular regional and political backgrounds enhanced her social unease related to her own identity:

I'm an [STEM] student so I'm around a lot of guys sometimes...but a lot of them are from rural communities. Like a lot of them are ag farmers which are from rural communities which are generally kind of conservative. And sometimes I'll hear something that they will say and it will kind of bother me.

These experiences highlight the multifaceted nature of the college campus context and how college students strategically navigate particular social groups and settings when they perceive the potential for prejudice or discrimination.

Taken together, these examples show how identity development is a particular type of privilege for LGBTQ college students dependent on the supportive nature of their campus context. The overwhelming majority of college student participants stressed the freedom they experienced to be themselves while at college and the wide variety of

resources they were able to utilize. It is important to note, however, that the campus context is not unidimensional, as there were LGBTQ students who described the challenge of navigating prejudice and discrimination within specific social circles. Despite these unique struggles, college students agree that the campus context is broadly conducive to identity development, which in turn can oftentimes conflict with the larger community's attitudes towards LGBTQ people and how underprivileged populations pursue identity formation.

### Homelessness as a Hindrance to Identity Development

While homeless young adults consider their LGBTQ identity to play an integral role in their lives and spend time considering its impact on their experiences, their social context of homelessness, which includes living on the streets, couch surfing (i.e., staying with multiple friends and/or family members for short periods of time) and residing in a number of different service agencies, acts as a hindrance to identity development as a whole. In sharp contrast to LGBTQ college students, LGBTQ homeless young adults must engage in concerted, deliberate efforts to explore their identity by seeking out resources and social support that are not as readily available as they tend to be on a college campus.

Homeless Experience Overrides Identity Development

Many LGBTQ homeless young adults stressed the importance of their identity at the same time that they acknowledged the distinctive challenges they faced in not having a regular place to live. However when considering how their LGBTQ identity shaped their experiences of homelessness, homeless young adults often described residential instability as a hindrance to developing their identity. In many ways, the search for basic

necessities such as food and shelter overrode young adults' ability to pursue identityrelated activities, such as self-reflection and social support. Abby (straight transgender woman) responded in the following way to describe how being homeless has affected her gender transition:

It's been an annoying, aggravating process because since I don't have a regular place to go to, like an actual residence, it kind of slows down the [gender] transition a bit because I don't always know when I'm going to get my next meal. I don't always know when I'll be able to get the certain necessities that I need.

Similarly, Sawyer (lesbian transgender man) recounted a similar experience in balancing the struggles of being homeless and lacking social support and simultaneously transitioning to his preferred gender:

It just made it harder when I was living on the streets and had nothing...Cause everything was just so difficult, every aspect of life. And that only made it harder, trying to deal with it [gender transition], especially when you're being quiet about it. I never felt so alone before in my life.

LGBTQ homeless young adults, therefore, express the desire to explore and understand their identities, but they are unable to fully engage with identity development when they are struggling to meet even their most basic needs.

Combined with the challenge of procuring food and shelter, some LGBTQ homeless young adults discussed the difficulty of even finding LGBTQ resources they could utilize when they were living on the street. Bernard (gay man) also downplayed his LGBTQ identity when describing how his sexuality was not at the forefront of his mind when he became homeless:

It was more being afraid of being on the streets, more of where I'm gonna go, where I'm gonna sleep, how am I gonna get food. So I didn't know any communities, I didn't know anything about [city]. So that was just the scary part of it...the LGBTQ part didn't come in until I got to [youth service agency] and started coming out to my caseworker and she was trying to find supportive environments for me to go to and [people to] talk to, cause I didn't know where

the LGBTQQ support group was when I became homeless...I had no phone, I had no way to contact anybody and everything was just gone.

Even when services may be available for LGBTQ homeless young adults, they often have more pressing survival matters to attend to and may be unaware of such services existing if they have no means of communication.

Other young adults also felt that being homeless made it especially difficult to form any type of romantic relationship, which was further exacerbated by the prejudice and discrimination they faced because of their LGBTQ identity. For Bianca (bisexual woman), residential instability acted as a formidable obstacle to establishing meaningful partnerships:

Well, it's kind of hard to get attached to somebody if you have to go from place to place to place. It's hard to just be with somebody, worried about somebody else when you don't even have yourself together. It's just more stress especially with a child and you just going from house to house just staying with anybody because you don't want you and your baby out in the street, you know. So that's hard and then on top of that trying to deal with a relationship. That doesn't do anything but make it 20 times harder. So, yeah, I'd rather just be single until I can get myself together, till I can get things stable for me and my daughter and then I can worry about being in a relationship. But until then it's not going to happen.

Bianca's experience of homelessness was further exacerbated by her status as a mother, whereby her caregiving responsibilities, combined with her lack of housing, took precedence over her LGBTQ identity. In this way, it can be especially difficult for LGBTQ young people to develop social support networks when they cannot secure any type of permanent housing, let alone access to resources and opportunities for LGBTQ matters.

Creation of Homeless Social Support

Despite the distinctive challenges LGBTQ homeless young adults face that prevent them from engaging with identity development, many respondents sought to overcome the barriers homelessness imposed upon them. One way homeless young adults coped with the struggles of not having a regular place to live was through the creation of social support networks they could draw from in times of need. Many young adults in this study experienced residential instability from a very young age and found ways to establish alternative networks of support beyond their nuclear and extended family. Yolanda (bisexual woman) found that treatment centers were significantly more welcoming than her home life, even though she acknowledged that this was not the norm:

It was more like a family, and I know that's not normal, but almost everywhere I went, there was like, I went to [name of treatment center] treatment facility and there was only like 2 females that were straight on my unit and all the others were bisexual or lesbian and I felt like I was home. I felt like I had a family there because we all knew that we were different but at the same time we were the same. So, strangely enough, at every treatment center I felt like home.

Another young adult, Lawrence (gay man), recounted his positive experiences in juvenile detention because of the social support he experienced:

It was better. It was better than being at home...I got attention. People that you could actually talk to, I mean yeah they were all criminals and they made mistakes, but at least I pretty much got a better place to live, it was better than nothing.

Despite the challenges these youth faced as they transitioned across multiple detainment situations, they were still able to create sources of social support that filled a void in their lives as many were detached from their families. Similarly, many of these young adults found ways to create LGBTQ support networks with other detained youth, which underscores the agency they exerted in developing their identities when little to no organizational support structures existed. It is important to note, however, that despite receiving and creating support when lacking family connectedness, these created support

networks were often with other deviant youth, which could affect their ability to exit homelessness (Auerswald and Eyre 2002).

In many ways, experiences of homelessness created a sense of social solidarity by uniting people through a shared sense of struggle and conflict. Melanie (bisexual woman) emphasized that fellow homeless people were accepting of her LGBTQ identity and did not discriminate on the basis of gender or sexual orientation:

I think people are a lot more open in homelessness about it than... it seems to be the more upper class you go the more like judgmental they get. So like being down at the bottom, people are a lot more open because they have been there. They've been through things. You find ways to cope and everybody is just like 'okay yeah, you are who you are.' So there's not near as much judging when you are down on the bottom than when you're up higher.

Melanie's description of elevated social class as a barrier to acceptance stands in contrast to some college students' descriptions of higher educational levels being linked to more inclusive, less biased attitudes, such as in the case of Natalie ("Like education tends to make you less homophobic, go figure.") Like Melanie, Elliott (bisexual man) shared a similar sentiment in how he has found support through other homeless people and experienced more challenges related to being bisexual than from being homeless. When asked how being homeless has affected challenges related to his identity, he replied:

I wouldn't say challenges for being homeless. There are challenges but not as many because people are in the same predicament that you are because you're homeless...I've met other transgenders [people] or other bisexual or transgender couples who are homeless and the homeless... despite what other people think, we do kind of stick together because we can protect each other that way.

These young adults' experiences demonstrate the importance of created kinship among homeless communities, particularly when support from service agencies is not always guaranteed.

Navigating Acceptance/Discrimination from Service Agencies

When LGBTQ homeless young adults were able to locate and utilize various services, their experiences were mixed in terms of perceiving a combination of acceptance and discrimination from local service agencies. The majority of homeless young people described their interactions with service agencies as generally positive, especially when they were geared towards youth and young adults and/or emphasized particular supports for LGBTQ individuals. Tamara (gay transgender woman), for example, stressed that the youth service agency she frequented made her feel like she fit in and acknowledged her uniqueness:

They treat me normal, like I'm a human being as well. But then again, they know how I am, so, but everybody, every transgender [person] has their own individuality and I believe they have their own individuality that separates them from other transgenders [people], which they do.

Many of these young people simply wanted to be able to utilize homeless services without fear of prejudice or discrimination at the same time that they value organizations that acknowledge LGBTQ identity. In stressing the importance of LGBTQ groups for homeless young people, Kellen (gay man) described these organizations as safe places for youth like himself:

Well here they're trying to make it more accepting for those who come in...to show that it's ok, like you don't have to hide it, you don't have to worry about it, other people hurting you or anything for it. And trying to teach those who aren't as accepting to be more accepting, I guess. And it does help, because if you're ever in danger or anything, you have this place you can go to.

As these examples show, homeless service agencies, and specifically those designed for LGBTQ young people, can be crucial sources of support for youth who are doubly marginalized and in need of assistance.

Other LGBTQ homeless young adults, however, noted that there were certain service agencies that they avoided based on their own negative experiences with these organizations. Oftentimes, service agencies that exhibited prejudice and discrimination towards LGBTQ individuals were religious-based or widely known for their conservative political views. In some cases, adult homeless shelters appeared to accept LGBTQ individuals, but placed more regulations on their behavior compared to non-LGBTQ residents. Harris (gay man) provides an example of this type of treatment that he experienced at a local shelter:

The shelters are kind of open to it, but not really. They don't treat people differently, but they kinda disregard a lot of the LGBTQ people. They don't really agree with us, but they are open to letting us stay, but they don't let us do what we want, basically. Like, I couldn't stay with my boyfriend because it made them uncomfortable, I guess.

Similarly, Bernard (gay man) expressed his concern in utilizing Catholic-based services because he felt that the workers viewed his gender presentation and his sexual orientation negatively:

I believe it was more of religious views...that was the issue. I've had a couple experiences at some donation places not being so supportive as well as homeless shelters...so it was just some of the community support system, such as the shelters and organizations like [Christian-based organization] wasn't as, didn't feel so supportive and accepting, I got a lot of judgmental looks and I just didn't feel like I was safe to go to those areas without other people to come with me, and that was during the time that I was walking the streets.

Even though these young adults were in dire need of institutional support and services, in some instances they viewed it as preferable to avoid service agencies altogether if they perceived them to be discriminatory.

Not all young adults' encounters with homeless service agencies could be characterized as simply positive or negative, as some LGBTQ young people altered their

views of particular agencies after interacting with them. When asked about his experiences with religious-oriented organizations, Elliott replied:

Okay, Catholics don't really like... they're not really against, they just don't agree with it. Actually asking for help at the church while being homeless and the way we were [LGBTQ], we weren't really comfortable going to other churches but we did end up going into a Catholic church and a pastor actually creates groups for the LGBTQ and now that he is retired from the church he does sponsorships and goes and does meetings for us, you know, for everybody. And after that I kind of, I made a friend of course.

This type of interaction underscores the complexity of homeless LGBTQ young adults' relationship with service agencies and the influence of cultural assumptions regarding the idea that religion is inherently hostile towards non-conforming sexual and gender identities. While Bernard felt explicitly discriminated against from religious-oriented groups because of his non-conforming gender presentation and sexuality, Elliott experienced a welcoming environment in a Catholic church, despite his preconception that all Catholics were prejudiced against LGBTQ people. These experiences demonstrate the complexity of homeless young people's encounters with service agencies, particularly when those organizations followed a religious ideology.

## **Summarizing Identity as Privilege**

Overall, LGBTQ homeless young adults' experiences uniquely differ from LGBTQ college students in terms of their ability to engage in processes of gender and sexual identity development. Compared to college students, homeless young adults face greater obstacles to establishing their identities and they responded to these challenges in a variety of ways. The homeless environment largely acts as a hindrance to identity formation, and LGBTQ young adults must engage in deliberate efforts to overcome the barriers posed by lacking a stable residence. Many homeless young adults found that the struggle to acquire basic necessities often trumped a concerted focus on their gender

and/or sexual identity. However, young adults coped with these identity-related challenges of homelessness by constructing networks of social support and strategically utilizing services that were supportive of LGBTQ people and avoiding those perceived to be prejudiced. While both LGBTQ college students and LGBTQ homeless young adults both sought to develop their identities, social contexts played a key role in shaping their access to supportive resources.

### STRATEGIC IDENTITY MANAGEMENT

Though LGBTQ college students and LGBTQ homeless young adults are exposed to widely differing social contexts, they engage in similar tactics to strategically manage their identities. These strategies, however, do tend to vary depending on the social environment in which the young people are immersed as they adapt to their social surroundings. Despite these differences, the ways LGBTQ young people navigate their family relationships acts as a link across social context that highlights the complex similarities between homeless and college young adults. Furthermore, both groups of LGBTQ young adults express a strong desire and determination to maintain control over their identities by deciding when and how disclosure occurs.

# **Identity Suppression**

One key way that both LGBTQ college and homeless young adults strategically manage their identities is through processes of identity suppression, which can operate both internally and externally. Young adults typically engaged in identity suppression when they perceived the potential for prejudice or discrimination and sought to avoid various forms of bullying or victimization. Oftentimes, young people recounted middle

school and high school as times that posed significant challenges for them in terms of social acceptance and how they suppressed their LGBTQ identity as a way of coping with unaccepting environments. Renee, a college student, succinctly summarized her middle school troubles in the following way (bisexual woman):

I was not happy. It was the unhappiest time of my life...I was just like frustrated because I had crushes on these girls and, you know, but also with guys and I didn't want to tell anyone. I didn't tell anyone and... yeah, it was just a really bad time keeping it all bottled up.

In a similar vein, Kiley (bisexual woman), a homeless young adult, described the difficulty she faced in attempting to come out in school and being rejected by her peers:

Well that's kind of when I started to come out more, not as much, but I tried to come out...cause my gay friend made it seem like I was trying to hit on all my friends, cause it was in 8<sup>th</sup> grade when I was actually realizing that I was officially a bisexual. But I didn't really come out then, I was trying to, but I didn't, because everyone looked at me like I was weird and disgusting, so I just played it off like, 'oh no, I'm not, I was just joking' type thing. It was hard though, because people would look at me funny, when they heard the rumor or whatever and I just kept telling them, 'no, it's not true, it's not true.'

According to these young people, having to repress their identities as a means of garnering social acceptance was a distressing experience, especially while in middle school when they are already facing numerous other challenges typical in adolescence.

While their experiences of identity suppression were similar in their earlier lives, social context emerged as a key component in shaping how LGBTQ young people concealed their identity as they transitioned into adulthood. For LGBTQ homeless young adults, the homeless environment posed unique challenges, often forcing the youth to suppress their identity as a way of staying safe from harm. This strategy typically involved trying to pass as straight or cisgender in environments they viewed as threatening. Harris's words summarize the extreme danger some LGBTQ youth face on the streets: "It's tough, because you can't really come out because there are some people

that don't agree with it and they will kill you because of it." In describing how he manages his bisexual identity on the street to stay safe, Patrick (bisexual man) replied:

I'm pretty secretive about it. As in, I don't really assert myself as bi, I would assert myself as straight. More so, because it's more for my protection from people in the community that would want to hurt me...one time somebody had thought something about me and I had to literally defend myself, so, I don't like to have to fight over my identity.

Similarly, Kellen recounted how it was difficult to always know who would be accepting, so it was often safer to act straight:

I just had to be a lot more cautious because um, I don't know, you have a mixed homeless community, the homeless youth community is a mix of everybody from everywhere, and not everybody in some places are accepting...I guess you gotta act different than you would normally...more like you're straight, I guess you could say...pretended that you were interested in women or whatnot.

For these homeless young people, LGBTQ identity suppression can be a very real survival tactic that can mean the difference between safety and victimization.

While LGBTQ college students also faced threats to their identity on the college campus that led them to conceal their true selves, their experiences differed in that they feared distinctive types of social alienation and rejection, rather than outright physical victimization. In many ways, LGBTQ college students sought to blend in with their desired social circles, rather than have their identity mark them as different. Bethany, for example, recalled how she navigated her identity in particular social groups so as to maintain heteronormative standards:

Sometimes, like in certain situations, like girls, my sorority sisters like to talk about like relationships and stuff and sometimes I'll bring up like something that happened in a relationship that I used to have and I'll change the pronouns of the person I was dating. Like I'll either just say 'they' or I'll change it and pretend they were a 'he' even though they weren't. Sorority is kind of like a small town and if you tell one girl then all 90 other girls will find out. So I just don't want people to start looking at me differently until they've really gotten to know me and they know it doesn't make me anything different.

Related to classroom experiences, Carrie (bisexual woman) had to make a conscious effort to conceal her identity when asked to reveal more personal information about herself by an instructor:

There were a couple of moments where I like... I don't know, I had to answer questions differently with my professors and stuff because I didn't want that to be revealed or whatever. It's just hard to know. Some teachers I think are going to be more accepting than others and you just don't know which ones those are.

On the college campus, LGBTQ young adults must also engage in identity management by strategically identifying situations in which they felt more comfortable keeping their identity a secret.

From these LGBTQ college students' perspectives, maintaining stealth within their identities was a process they engaged in when the situational context demanded it, whereas LGBTQ homeless young adults described the effort of identity concealment as an omnipresent issue on which their safety hinged. LGBTQ identity appeared to surpass differing social contexts for both groups during the years of middle and high school, as all of the young adult participants described middle school in particular as a time of heightened social and personal distress. While LGBTQ young adults manage their identities in response to their wider social contexts, their identity strategies become more nuanced and complex within their personal and professional relationships.

# **Navigating Identity in Social Relationships**

LGBTQ young adults must utilize context-specific strategies to manage their identities as they transition between different environments, and they further refine and tailor these strategies as they navigate their identity in their multiple personal and professional relationships. Most notably, both college and homeless young adults faced challenges in determining how to disclose their identities to various family members

within their nuclear and extended networks. Harper (lesbian transgender woman), a college student, strategically determined a "need-to-know" system of disclosure within her family, as she stated:

I only came out to my best friend at the time, my brother, my mom, and then I haven't come out to anyone else because those are sort of the people that I've decided... that's my family, that's my immediate family unit it was very much of a fear of mine about sort of rejection so I only shared it with people that I had to share it with because I used them as confidants. I've done my part and the ball is in everyone else's court if they want to know because it's not my job to stop their bigotry.

In differentiating between her nuclear and extended family's knowledge of her identity, Harper further explains:

And like with my other family members...I could come out to them but that wouldn't do anything. That would just cause a lot of friction. It would just be something else to fight about, something fresh...I mean like all the old wounds are already scarred over with those relationships so they can be talked over and they don't have to be brought up but introducing something new could potentially sort of hamper that piece.

In this detailed explanation of managing disclosure within her family, Harper represents the complex ways that LGBTQ young adults engage in a type of cost-benefit analysis of identity disclosure.

LGBTQ young adults also described contending with the management of the flow of information related to their identity across different family members. For Julian (bisexual man), a homeless young adult, choosing not to tell certain people in his family afforded him the ability to control which family members knew about his identity:

Step-dad, I really don't care about him, but he's a homophobe anyways. That's one of the main things that's holding me back from telling him. He and I really don't like each other. If I were to come out and tell that and he doesn't know what happens in the family stays in the family, that sort of privacy thing, I feel he's going to tell everybody in [city] and beyond. I like my privacy and that's not just something I want everybody knowing.

Relatedly, many young adults also had to navigate the role of familial gatekeepers within their family who expressed a desire to disclose the young person's identity on their behalf, either with or without their permission. One college student, Sophie (bisexual woman), experienced forceful gatekeeping behavior from her sister, as she described in the following way:

She told me that it was important that I tell my parents. It was important to her. She was putting on me the fact that by not telling them, I was lying to them and that was bad and whatever. I was like, 'You know what? When it's the right time, I'll tell them. It's not really your place to tell me that now is the time.' After that, she unfriended me on Facebook and stopped talking to me. Ever since then, it's been strained between her and I. I don't think that she's told anybody else in my family. She's the only person in my family that knows.

From both Julian's and Sophie's accounts, it is clear that navigating identity and its disclosure within the family is an important aspect of LGBTQ young adults' lives in its influential role in family dynamics.

Friendship relationships also posed unique challenges for LGBTQ that were similar to, yet distinct from, their experiences with family members in regards to managing their identity. Melanie, a homeless young adult, experienced a form of gatekeeping from a close friend in middle school that was characterized by a mixture of support and disregard for privacy:

The biggest thing I remember is telling my best friend at the time, who was completely gay, how I felt and him supporting me a lot, but he had a really big mouth so he went and told everybody and then everybody kind of came at me at the same time but nobody really condemned me for it. They were just all like 'oh hey, cool.'

College students also used distinctive strategies in revealing their identities within friendships, such as Jake, who engages in a type of "pre-screening" when determining to which friends he discloses his identity:

It's been to my friends who are pretty progressive by and large. You don't want to come out to anyone who would think less of you or judge you. Not like I give a shit what they think but you don't want to negatively impact what could be a fruitful relationship beforehand if it might not necessarily be so after. I don't know...just the people that I am closest with are the people who I think would accept me unconditionally.

It is clear from these examples that identity poses similar barriers for LGBTQ young people, regardless of social context, as they negotiate their identity within personal relationships.

Though the majority of LGBTQ young adults emphasized their family and friend relationships when discussing how identity shapes their social worlds, a few participants elected to describe how their gender and sexual identities interacted with the workplace. Professional relationships can be especially important for young people as they work to develop their career trajectories, and the threat of prejudice and discrimination in the workplace is very real for LGBTQ young adults. In describing his coworkers' minimization of his transgender identity, Gavin (bisexual transgender man), a college student, expressed his frustration in the following way when he was not able to easily access his preferred style of work uniform:

I really only just recently even got the guy's uniform because the woman that is in charge of ordering the uniforms and everything, I would tell her repeatedly 'I want a guy's uniform, please order me one.' Like it's not that hard to order a uniform because you order them every month and every time she would refuse. She always will like use nicknames. Like she'll call me 'mama' which super frustrates me.

Homeless young adults also faced issues in the workplace when they interacted with prejudiced individuals, and the example of Bernard responding to homophobic customers highlights how young people can assert themselves in these situations:

Usually it's telling them to leave the hotel...we're very open to the whole community, not just including the LGBTQQ, but more into even the homelessness type of community as well. So we're very non-judgmental and for

the safety of our guests and safety of our employees...I don't tolerate any guest coming in and making fun of me, well I'll allow them to make fun of me, I'll allow them to call me a fag, but anything to my other employees is something I'm very protective of and don't tolerate at all and I do ask them to leave the hotel.

Though these young people faced blatant prejudice and discrimination at their places of employment, the experiences of Gavin and Bernard show how young adults challenge homophobic reactions to their identity, and this type of agency is further expressed in how they conceptualize the role that identity plays in their lives.

### **Agency within Identity**

LGBTQ young adults face unique challenges in how they navigate their social realities, but this is not to say that they are simply at the mercy of prejudiced and discriminatory outside forces. In many cases, both homeless and college LGBTQ young adults actively confronted obstacles related to their identity by taking charge of situations and exerting agency surrounding their identity, such as within disclosure processes. Despite facing tangible obstacles to identity formation in the forms of prejudice and discrimination, these young people sought to establish control over if, when and how they disclosed their identity, thus highlighting their sense of responsibility over their desired life course trajectory and their agentic motivations (Côté and Levine 2002). One homeless young adult, Felicity (lesbian woman), bluntly described her attitude towards selective disclosure in the following way: "I just, you know, I feel like if I am gonna come out to someone, it should be my privilege to tell them and if I don't wanna tell them, I'm not gonna tell them." In the classroom context, Natalie, a college student, also exerted agency in how she mediates her LGBTQ identity through her presentation of self that she uses to draw from her privileged social statuses:

But for me nobody knows that I'm anything other than a heterosexual straight cisgender white girl unless I tell them. So I have a little more control in that way

and I will kind of gauge by the class whether or not it's appropriate to mention that. And I won't ever bring it up if it's not relevant at all.

These agency-laden approaches to disclosing (or not) their LGBTQ identity highlights how young adults sought to establish control within their lives, rather than act as passive observers in a society that continues to marginalize non-conforming genders and sexualities.

Several young people also described agentic processes that were more heavily tempered by structural constraints, such as perceptions of negativity that encouraged them to conceal or minimize their identities. As another means of establishing a sense of agency throughout processes of disclosure, other LGBTQ young adults attempted to downplay or minimize their own gender and/or sexual identity in relation to their other roles and identities. Charlie, a college student, represents this strategy in how she asserted that she does not allow her LGBTQ identity to define her:

So I guess I don't ever give it the opportunity to get to my relationships. I try and keep it, not like a hidden thing, but more of a 'I'm more than my sexuality and gender identity' kind of thing but also I mean I do hide it in a sense at the same time.

Similarly, another college student, Billy (gay man), argued that discovering his identity in middle school was par for the course and not at the forefront of his thought processes:

I don't really think it had any big bearing on it. I mean I don't really see it as having had much of an effect then because it wasn't really like something I gave too much thought towards. Because it was kind of one of those things I came to the realization of while I was going about my regular business, hanging out with my regular friends and everything. I didn't feel like it changed anything at least.

By relegating their gender and sexual identity to a status that does not define who they are, LGBTQ college students exerted agency in their lives and social relationships.

Homeless young adults also moderated the power they allowed their LGBTQ identities to have over their lives, but in many ways their outlooks tended to take on a

pessimistic tone rather than an empowering mentality that characterized the college students' descriptions of agency. When asked to describe the best thing about being LGBTQ, Yolanda was unable to identify a single positive aspect related to her identity, and instead replied, "I don't think there is a best thing, sadly. Cause no matter where you turn, there is some type of judgment. I mean, it's not really a gift, but it's not really a curse either. It's just life." In a related vein, Jackson (bisexual man), who was temporarily staying in a transitional living house, attributed his challenges in life to lacking a regular place to live rather than his bisexuality:

I think the challenge isn't my identity. It's the homeless part. I don't really see too much ... too many challenges that come with being bisexual, other than people won't shut the hell up about it. But being homeless is a little bit different because you don't have a permanent place to go home to. I mean, yes, I do have shelter over my head but it's only going to be for about another month or two.

In contrast to their college counterparts, LGBTQ homeless young adults framed their agency in a much more resigned way by downplaying their LGBTQ identity, which most likely relates to the saliency of the struggles they face within homelessness.

### **Summarizing Strategic Identity Management**

The types of strategies and agency LGBTQ young people exhibit as they manage the impact of identity within their lives speaks to the ability of young adults to frame experiences in ways that are advantageous to their social relationships and contexts. While both college students and homeless young adults found ways to exert agency within their identities, social context continued to shape their perceptions as college students adopted an empowered framework, while homeless young adults displayed pragmatic approaches to their LGBTQ identity. These experiences of agency are indicative of a broader pattern of resiliency that emerged among participants and transcended social context. Resiliency is defined here as the ability of young people to

positively adapt to conflict-laden life experiences, such as sexual and gender identity-related struggles, whereby protective factors, including self-confidence, work to promote successful development in the face of adversity (Russell 2005).

#### IDENTITY AS LIFE ENHANCEMENT

Both LGBTQ college students and homeless young adults face distinctive challenges and experiences related to their gender and sexual identities, and social context often plays an integral role in how young adults perceive their lives. One aspect of LGBTQ young adults' life courses that appears to supersede differences across social context is that of identity acting as a form of resiliency and how it enhances their lives in meaningful ways. For example, LGBTQ young adults generally described their identity as a positive aspect of their lives that helped them navigate different types of relationships at the same time that it developed their sense of themselves in relation to other people and social issues. Overall, these young people conceptualized conflict as a type of resiliency that helped to shape them into stronger, more capable individuals.

## Identity as a Relationship Modifier

In many ways, LGBTQ identity provided a means of modifying relationships for young people that helped them to assess the value of connections they had with various people in their lives. For the majority of LGBTQ young adults, their identity helped them build up and strengthen relationships with particular people. Holly (lesbian woman), a college student, believed that her identity provided more opportunities for friendships as she welcomed opportunities to act as confidant for others who were struggling with their sexuality:

I think that it opens up a lot more doors for, not just for conversation, but even with females that are confused. They come to you more. They disclose a lot more

personal information...same with males. Anyone confused about their sexuality, I think that opens up opportunities for friendship because you like to confide in people that have similar situations so I think it builds stronger relationships.

In a similar way, Harris, a homeless young adult, was confident that his gay identity helped him to form relationships with people by opening lines of communication:

It actually made it a lot easier because then it's helping others to realize things easier once you get to know a lot of them that's been out there a lot longer. It's helped open up relationships a lot more... because now that others realize who I am, truly, it makes it easier for them to open [up] and it's easier for me to get closer to them.

By coming to terms with their identities, these young adults felt that they were able to build deeper relationships with people that were more fulfilling and authentic compared to social interactions where they remain closeted.

While identity could potentially create pathways to new relationships and solidify and strengthen those already established, LGBTQ young adults also perceived their identity as a type of relationship filter that revealed the true nature of their friendships and family ties. Sawyer, for example, a homeless young adult, believed that his non-normative gender and sexual identities granted him the ability to see people's true colors, as he describes in the following way:

It's made me be able to look at people, I can see through people now. I'm able to read people really well now, having that experience of people coming and going often and being really fake towards you. It's like I'm able to read that well now...it's made me realize the people who truly care, you know they can see past it. Cause I've lost friendships because of it, you know just people not accepting of it.

College students also expressed similar sentiments, such as in the case of Gabriel, who claimed that his identity allowed him to be more assertive in his social relationships:

I don't really put up with people who aren't accepting of me and I don't allow people to walk all over me so I think that it really helps me build more meaningful relationships. So I feel closer to everybody. I may not have thousands of friends, but the handful of people that I do have, I actually know that they're going to be there for me

In essence, these young adults valued the quality of their relationships over their quantity, and attributed their ability to form these meaningful social connections to their LGBTQ identity.

# **Developing Social Awareness and Empathy**

In addition to their identity opening doors for new and more meaningful relationships, many LGBTQ young adults also concurred that their identity provided them with experiences that helped them develop their own social awareness and sense of empathy towards others. As a homeless young adult, Sonya (bisexual woman), was deeply aware of others' suffering and she asserted that a bisexual identity enhances one's empathetic nature:

You love everybody. I feel like there's no hate towards another race. Like... as a metaphor, towards another person. Like we have bigger hearts than anybody else, because you can love somebody even knowing that somebody else doesn't like you for loving that person. Or you can just be who you are.

Heidi (asexual woman), a college student, believed that her identity shaped her ability to be more open to different kinds of people. When asked how her identity affected her social relationships, she replied:

I think it makes me a little more accepting of other people and of their experiences. I try to be less likely to assume that somebody in the street was a certain orientation...depending upon if I just looking at them. It's a process but I try not to do that. I try to use more inclusive language. I would say I'm more open and accepting of other ideas...I'm a "live and let live" kind of person.

These introspective processes stemming from the young person's LGBTQ identity developed their social awareness to the point that it extended into their social interactions and relationships as well.

Beyond a level of heightened introspection and awareness regarding social issues, LGBTQ identity also prompted young people to be more outspoken in their beliefs as they modified their social interactions. Carrie, a college student, recounted the shift in her reactions to prejudiced behavior after she came to identify as bisexual:

I definitely get offended more easily if somebody, I don't know, makes a joke or something that's kind of inappropriate...before I would kind of just shrug it off like it wasn't a thing and now I'm much more apt to like call them out on it and yell at them kind of thing. And that's not just in regards to sexuality too. I think it's just because I feel like I'm a member of a minority that I'm much more sensitive to people of other minority groups as well.

Furthermore, as a result of experiencing prejudice in her own life as a homeless young adult, Willow (lesbian woman) recognized the importance of accepting others and made it a point to encourage those around her to do the same:

I told everyone like life, well life can be easy, it all depends on how you like control your life...life is [too] short to judge anyone, like nobody has the right to judge anyone about how they feel or what sex they like. Judging is a big word, because if you don't know exactly what the person is about, you don't need to judge them for where they come from. If you understand them, then you understand them. But if you don't know anything about them, don't judge a person just because of how you feel about it, you never know what the person may be going through.

By sharing their beliefs regarding identity and social acceptance, these young people used their gender and sexual identities as a way to stand up for their ideals, which had additional benefits for their mental health.

## **Identity as a Boost to Personal Well-being**

As LGBTQ young adults described how their identity had impacted their lives and experiences, their stories touched upon a number of different aspects, ranging from the social and interactional to the more personal. One of the more intimate experiences they recounted was how coming out as their gender and/or sexual identity acted as a major boost to their sense of personal well-being. Thomas (gay man), a homeless young

adult, revealed that his LGBTQ identity helped to boost his level of confidence in himself, as well as his sense of independence:

I just confidentially knew who I was, so it was like, it was natural. I didn't have to go through no transitional like state or emotion. I just love who I am and fuck what anyone tells me is how I feel about it... But I'm there for myself now. That's the thing, because I don't wanna put my strength into other people. You have to be by your own side. That's confidence. That's what works.

In a similar way, Sophie, a college student, noted that confidence stemming from her bisexuality helped her to accept herself and build stronger relationships:

I think that generally, ever since I was honest with myself about being bi, I feel a lot more confident... When you feel like you have that secret and you're embarrassed by it, or that no one can know, it really takes a toll on your self-esteem. I think it's caused me to grow as a person, in a lot of different ways... it's given me a lot more confidence and I feel like people can see me for who I am.

By conceptualizing their LGBTQ identity as a source of confidence, these young people felt that they could be more true to themselves and comfortable in their own skin, which also imbued them with more pride in their identities and improved their overall sense of self-esteem.

In addition to equating LGBTQ identity as a confidence-booster, many young adults also described coming to terms with their gender and sexuality, on both personal and interactional levels, as freeing and liberating. When asked to describe how being queer has shaped her life, Clark, a college student, replied in the following way:

So I think it helps me be flexible because queer doesn't mean like I have to be only attracted to women always and I can't have any sort of desires or feelings outside of that. And so queer gives me the space to be able to be who I am... I think it makes it easier to be freer from shame.

Relatedly, Abby, a homeless young adult, attributed her mental clarity to the fact that she was transitioning and realizing her true self:

Life has actually become more clear. I've always wondered how my life was going to be like after transitioning and soon as I started and up to now it's become

a lot easier for me. It's become, so far, in most ways a better life for me. Everything I've seen myself doing in the past it was a little blurry back then but now it's much clearer because I'm actually taking action.

For both college students and homeless young adults, an LGBTQ identity has the potential to enhance their sense of purpose and meaning in life when they have the freedom to fully be themselves.

The struggle for authenticity also emerged as an important factor related to LGBTQ young adults' personal well-being. While identity as source of confidence and liberation were widely discussed as positive elements related to their gender and sexuality, a sense of authenticity was described as a more elusive goal for young people. In the following way, Carrie, a college student, explained the ambivalence she experienced when trying to balance her sense of authenticity and her personal privacy as she managed disclosing her bisexuality:

I'm a pretty open and honest person and I want to be honest about all of the aspects of my life but at the same time I don't think I should have to share something about myself that is so private...so it's a lot of conflict between wanting to be completely genuine and also like being respectful of my own privacy.

Lawrence, a homeless young adult, also endured conflict as he struggled with "lying" to his family by concealing his gay identity. When asked how this affects his family interactions, Lawrence replied, "It hurts. I don't like lying. I'm not a lying kind of person, I'm the most honest person that anybody could meet, but it's very difficult to lie to myself and lie to my grandparents, lie to my whole family...but pretty much I'm just saving their feelings." Though young adults like Natalie, a college student, described "authenticity" as one of the best things about being LGBTQ because "it's people who know who they are and are cool with that no matter what other people are cool with," the

majority of young people found being true to themselves and others was one of the major challenges they faced.

### **Conflict as Resiliency**

While LGBTQ identity symbolized numerous positive traits for young people, they still endured various challenges in their lives related to both their gender and sexual orientation, as well as their particular social context. All of the LGBTQ young adults in this study, inclusive of both college students and homeless young adults, recounted enduring some type of identity-related conflict, such as prejudice or discrimination. These experiences certainly impacted their lives and well-being, yet, according to the young people, these challenges were formative in that they helped them become stronger, more resilient individuals. Conceptualizing conflict as resiliency cut across social context, yet young adults' perceptions of the most salient *source* of their resiliency differed for college and homeless young people.

For both groups of LGBTQ young adults, conflict related to their gender or sexual identity helped them build up their own personal sense of strength. Kiley, a homeless young adult, interpreted the prejudice and discrimination she faced as a bisexual woman as positive life experiences:

I guess one of the things that I got out of it was becoming a stronger person because of being what I am, I can embrace a lot of hateful things from people and practically not let it get to me because you get a lot of crap when you're bisexual...you gotta put up with a lot of crap from people and I guess I just became a stronger person because of all the crap I've had to put up with.

Similarly, Paige (lesbian woman), a college student, believed that having an identity challenge to cope with in life is "healthy" and improves her social relationships:

I just think it's healthy to have had something, I mean everyone deals with something, but I think it's good to have something that is difficult in your life just because I think it keeps you from getting too complacent about things. And I

think that it makes you appreciate when you are able to be yourself more...I think it just makes me value my relationships more, because there have been challenges to all of them so I think the people who are still around are probably gonna still be around and it's nice to know that you can get through some things with them.

Overall, gender and sexual identity-related conflict as a type of resiliency-builder cuts across social context for LGBTQ young adults, yet resiliency stemming from conflict within one's environment distinguishes college students from homeless young adults.

While the idea of conflict as resiliency is important for both groups, LGBTQ homeless young adults underwent unique challenges related to their lack of permanent housing that was not shared among college students. Yolanda, for example, adamantly believed that her difficult experiences with foster care, being homeless and self-harm molded her into a more resilient young adult:

I wouldn't change any of the experiences I've been through because without all of those experiences I wouldn't be as tough emotionally, physically and mentally. I wouldn't be able to say, 'I've been through, I lived that, I'm still here.' I mean, a lot of people they ask, 'are you ashamed of the scars from your self-harming?' I'm not. They're scars of survival and war to me. They're the story of my life. Cause I want to help youth in the foster system and make them understand that no matter what they go through, they can do whatever they want and be whoever they want. So, I mean, if I hadn't gone through half the stuff I went through, I don't think I would be as bubbly, as strong, as wise, as tough exterior-type person. I think I would still be that weak, shy, not able to stand up for myself, person.

Similar to Yolanda, Bernard also experienced challenges related to both being homeless and LGBTQ, which became most salient to him when his mother refused to provide any aid to him when he first became homeless and sought out her help. Despite enduring parental rejection and being forced to cope on his own without familial assistance, Bernard described being grateful for this arduous experience as it strengthened his resolve and his relationship with his mother:

It made me love my mom so much more because it built who I was today to I got out of homelessness on my own. My mom purposely did that because she wanted to see if I could do it on my own, which she said I would feel so much better

doing it on my own...and I do respect her for it...saying no, and declining helping me at all. It built myself back up, matured myself, and got me to the position I am in today.

Homeless young adults' perceptions of their residential instability as a source of resiliency strongly contrasts with college students, as students did not attribute any challenges in their life to the campus context, but rather to their LGBTQ identity broadly.

## **Summarizing Identity as Life Enhancement**

In considering the impact of their gender and sexuality in their lives, LGBTQ young adults generally perceived their LGBTQ identity to be a positive influence that enhanced their well-being in relationships in multiple different ways. Furthermore, these favorable views of how their identity shaped their lives was similar for both college students and homeless young adults, which highlights distinctive similarities across social contexts. Though both groups construed LGBTQ identity-related challenges they faced as a source of resiliency that made them stronger, more competent people, homeless young adults often brought their homeless identity to the fore of their experiences as an integral component of their sense of resiliency.

#### **CHAPTER 6**

### **DISCUSSION**

Findings here underscore the complex roles that gender and sexuality play in the lives of LGBTQ college students and homeless young adults. Both groups of young adults shared experiences that emphasize the energy and effort they exerted in conceptualizing and managing their identities on multiple levels, including the individual, the interpersonal and the contextual or community. The social contexts of the college campus and homeless environment deeply influenced how LGBTQ young people could engage with their sexuality and gender, resulting in identity acting as a type of privilege. Furthermore, young people strategically managed their identities depending on their encounters with different people and social contexts. Finally, LGBTQ young people often conceptualized their identity and context-related challenges as positive life enhancements by improving both their personal coping abilities and their social relationships. The following section highlights the implications associated with each of these themes and how they extend upon the theoretical frameworks of life course and emerging adulthood.

### **Implications of Identity as Privilege**

The theme of "Identity as Privilege" demonstrates the importance of social context within the lives of LGBTQ young adults and how these young people must navigate their environments in distinct ways when seeking out support and managing identity-related challenges. In comparing college students and homeless young adults, the campus context is largely conducive to identity development, as students are afforded a breadth of resources and opportunities they can utilize to construct their identities. Homelessness, on the other hand, overwhelmingly acts as a hindrance to identity

formation for LGBTQ young adults, and they must make concerted efforts to overcome these barriers and pursue identity development. In many ways, homeless young adults' experiences reflected a version of Maslow's hierarchy of needs, whereby they had to satisfy their more immediate needs of food, shelter and clothing before they could expand into self-actualizing identity work or self-transcendence through social engagement that was more readily available to college students (Koltko-Rivera 2006). This stark contrast between these two groups of young adults points to identity as a mechanism of privilege that is largely shaped by one's social context.

Despite the ways that social context created disparate experiences of identity development among LGBTQ young adults, similarities were also evident in how respondents viewed sexual and gender identity as an important aspect of their lives. For both LGBTQ college students and homeless young adults, social surroundings were key in determining how they expressed their identities openly or sought to conceal their identities if they perceived a sense of prejudice directed at their sexuality or gender presentation. Though social context acted as a source of privilege for college students in accessing support on campus, they remained aware of environments where the potential for prejudice existed, such as Greek communities. While college students perceive widespread campus support, they must still navigate the prevalence of heteronormativity that abounds in particular college contexts (Wickens and Sandlin 2010). Similarly for homeless young adults, social context shaped how they navigated different service providers at the same time that it prompted them to create social support networks in the absence of formal resources. In these ways, LGBTQ identity acted as a link across social

contexts that created both similarities and differences for college and homeless young adults.

A life course perspective considers the integral role of social context, as well as the interplay among individuals' social relationships and interacting identities (Elder 1998), especially the unique influence of gender and sexual orientations. My study demonstrated how LGBTQ young adults' social environments created distinctive social convoys for their transitions from adolescence to adulthood (Moen and Hernandez 2009). Findings here demonstrate the key role of young people's conceptions of their sexual and gender identities in their pathways to adulthood and how this transformative period is shaped by shifts in identity development (Torkelson 2012) and sources of inequality (O'Rand 2006). Social context perpetuated inequalities between college students and young adults, as college students experienced more ease and support in accessing resources, while homeless young adults faced many barriers in seeking assistance for identity development. Furthermore, this study expanded the scope of emerging adulthood to encompass a wider array of young people's life courses and their contexts, including LGBTQ young adults in college and those who are homeless, by emphasizing the empirical impact of the college environment (Arnett 2015) and that of homelessness (Ream and Forge 2014).

While this study captured a snapshot of LGBTQ young adults' lives at one point in time, it is also necessary to consider how their lives can shift as they enter into later life stages and experience formative life course transitions, such as graduating from college for students and establishing a stable residence for homeless young adults.

Graduating from the uniquely protective environments of college campuses could act as a

pivotal turning point for some students if they transition into contexts that lack support and affirmation, such as the workplace where they have no legal protections for their identities (Ragins and Cornwell 2001). Homeless young adults could also lose vital support networks if they had been utilizing LGBTQ homeless youth resources and age out, secure stable housing or otherwise become ineligible for services (Maccio and Ferguson 2016). Additionally, it is important to consider the fluidity of sexual and gender identities and how LGBTQ young adults may or may not be able to sustain their identities in changing environments if they lose key sources of social and cultural capital (Pinto, Melendez and Spector 2008). For example, sexual and gender identity can shift across the life course as individuals experience changing social roles, relationships and self-concepts (Diamond and Butterworth 2008).

Differences across race and class also emerged when comparing LGBTQ young adults in distinctive social contexts. For example, half of LGBTQ homeless young adults identified as people of color, while only 12 percent of LGBTQ college students were non-white, highlighting the intersections of race and class. The LGBTQ young adults of color in this study did not often mention the influence of race/ethnicity in their lives, however its role can be instrumental in the well-being of homeless young adults in particular, as racial and homelessness-related stigma can doubly harm their mental health (Gattis and Larson 2016), creating an intersecting matrix of systems of oppression that adversely affect marginalized people (Collins 2009). Similarly for college students, racial microaggressions are a detriment to self-esteem (Nadal, Wong, Griffin, Davidoff and Sriken 2014), which can further interact with on-campus homophobic slurs (Nadal et al. 2011) for LGBTQ college students of color. Furthermore, more college students (42%)

than homeless young adults (5%) identified outside of the LGBT spectrum, such as queer or asexual, which could point to social class differences in having more opportunities to learn about and educate oneself about non-mainstream sexual and gender identities.

Access to technology and informational resources are critical determinants in how LGBTQ young people can learn about their identity and issues related to their well-being (DeHaan, Kuper, Magee, Bigelow and Mustanski 2013). From this perspective, the college environment may not only be more conducive to identity development, but it can also expose college students to a wider variety of identity options from which they may choose, while homeless young adults may not reach that level of identity complexity as they struggle to meet their basic needs.

The notion of "Identity as Privilege" underscores the prominent role of social context in LGBTQ young adults' life courses and their journey through emerging adulthood. By acting as a source of privilege and as a barrier to resources and opportunities for different groups of LGBTQ young adults, social environment can help determine processes of identity formation. Furthermore, it is essential to understand LGBTQ young adults not as a monolithic social group, but rather as a dynamic, complex subpopulation that can simultaneously undergo similar, yet disparate, experiences that can both unite them and create distinctive intragroup differences.

# Implications for Strategic Identity Management

The second theme, "Strategic Identity Management," emerged as an important aspect in the lives of LGBTQ young adults that largely transcended social context as both college students and homeless young adults grappled with navigating their identities within their social networks. Social context continued to shape the strategies they used,

however, in that particular campus or homeless environments presented distinctive challenges to which the young people adapted in unique ways. In these ways, strategic identity management was a recurring concern for these young people as they were continually assessing their social surroundings and social relationships to find ways to protect themselves from prejudice and discrimination. Furthermore, young people were dedicated to maintaining control over their identities as a way to maximize the benefits of disclosing their identities or minimizing the risk of negative repercussions of disclosure.

For example, while LGBTQ college students suppressed their identities as a way to avoid bullying in school settings, homeless young adults often adopted stealth strategies regarding their sexuality and/or gender as a way to avoid street victimization, which could include extreme physical altercations. Participants in this study underscored the prevalence of LGBTQ-centered microaggressions present in young people's lives and the mental effort required to constantly be aware of subtle prejudice and discrimination in social interactions, including dirty looks and offensive humor. Microaggressions can vary across social context, however, as well as in their severity. Research shows the disproportionate levels of bullying and verbal harassment that LGBTQ college students face in campus settings (Woodford et al. 2014), however LGBTQ homeless young adults typically contend with much higher rates of physical and sexual victimization while on the streets (Cochran et al. 2002; Tyler 2008). Therefore, enacting a strategy of secrecy can be a very real survival tactic concerning their physical safety for LGBTQ homeless young adults, while for college students avoiding harassment was more conducive to their mental health.

The ways in which LGBTQ young adults manage their identities and navigate them within social contexts has vast implications for their mental health and social functioning. For example, young adults in this study faced varying levels of acceptance and rejection from their family, friends, peers and, in some cases, coworkers. Fear of discrimination and rejection was a primary motivating factor for both homeless and college young adults in determining if, when and how they disclosed their identity to their social networks. Research has highlighted the extreme importance of social acceptance in LGBTQ young people's lives, and how acceptance, or lack thereof, of one's sexual and/or gender identity can directly determine their mental health, including levels of self-esteem and depression (Ryan et al. 2010). While social support in relationships is key for all LGBTQ young adults, it can be especially critical for homeless young adults as their family relationships tend to be characterized by greater conflict (Tyler 2006) and problematic family ties can be influential in determining the resources and support they receive that can curb high-risk behaviors such as substance use and delinquent behaviors (Milburn et al. 2012).

Related to the broader implications of maintaining mental health among LGBTQ young adults in managing their identities, it can also be critical for these young people to be able to exert agency and control over social perceptions of their gender and sexuality. Regardless of social context, both LGBTQ homeless young adults and college students in this study often placed a high premium on taking ownership of their identity through acting as their own gatekeeper to disclosure despite contextual constraints that limited their ability to fully take control of social interactions. In many cases, though, homeless young adults faced more formidable barriers to exerting agency compared to college

A sense of agency can be key in promoting positive mental health among sexual and gender minority youth, as being out to more supportive people in one's social network can buffer the adverse effects of identity-related stress for these young people (Wright and Perry 2006). Glassgold (2007, p. 47) succinctly summarized the meaning and importance of agency in the lives of sexual and gender minorities: "Agency is an attempt to live as authentically in the world as we can, given its restraints and limits; and it occurs when we commit to meeting each limit with resistance, then finding a creative solution to transcend that limit." Some participants in this study also found the solution of identity concealment to be a source of empowerment, rather than one that harmed their mental health. Empowering LGBTQ youth and young adults can be a powerful mechanism in improving their lives if they are given the tools to confront sexuality and gender-related prejudice and discrimination (Craig, Tucker and Wagner 2008).

Processes of strategic identity management further extend upon understandings of LGBTQ young people's experiences through a life course perspective and using the concept of emerging adulthood. Particularly in considering young people's social networks, or "linked lives" (Elder 1994), LGBTQ young adults in this study emphasized the influential nature of social support or rejection that was most salient in the developmental periods of middle and high school. In this way, the intricate dynamics of family and peer relationships and the shifting quality of these relationships across the life course are key in determining how well they are able to cope with life's challenges (Moen and Hernandez 2009). Similar to the LGBTQ young adults in this study who found that establishing social solidarity with similar others was a critical source of

support (across identity and social context), other studies affirm this finding as they show that lesbian, gay and bisexual youth found the most sexuality-related support from other sexual minority youth (Doty, Willoughby, Lindahl and Malik 2010). Positive resources of social support are especially crucial for young adults as they attempt to successfully navigate the complex pathway to adulthood (Spencer and Patrick 2009).

### Implications for Identity as Life Enhancement

The third and final theme, "Identity as Life Enhancement," underscored the LGBTQ young adults' life course experiences, and social context acted as both a unifying characteristic and one that created divergent perspectives of personal strength. By conceptualizing challenges and conflicts as experiences that imbued them with resiliency, the young people in this study discussed how identity-related difficulties helped to improve their lives by strengthening supportive relationships, developing their social awareness and enhancing their overall personal sense of well-being. All too often, research over-emphasizes the risks and challenges that LGBTQ young people face, which can mask the potential these youth embody for experiencing healthy, well-adjusted development (Russell 2005).

Young people's capacity for framing adverse life struggles as positive, resiliency-building experiences can potentially improve their mental health and promote healthy coping strategies (Fergus and Zimmerman 2005). Particularly for LGBTQ young adults, establishing a sense of resiliency in response to challenging life events can help them manage prejudice and discrimination directed at their gender and/or sexual identity. For example, respondents in this study emphasized the importance of being proud of their experiences and establishing self-confidence related to their sexuality and/or gender. This

finding is in line with previous research that shows the complex ways that LGBT youth navigate homophobia through various behaviors, such as the development of a sense of mature responsibility and pride-filled identities (McDermott, Roen and Scourfield 2008).

While both homeless and college young people framed life challenges as positive experiences, some respondents' perceptions of the most influential source of resiliency in their lives was closely tied to their primary social context. For example, several LGBTQ homeless young adults constructed their sense of resiliency from their experiences with residential instability and homelessness. Research highlights the fact that resiliency among homeless youth can be especially important in curbing dangerous lifestyle behaviors such as drug use and suicide ideation (Rew, Taylor-Seehafer, Thomas and Yockey 2001). Furthermore, in light of the extreme struggles for survival they face on a daily basis, resiliency among homeless youth can foster enhanced self-esteem and self-confidence (Kidd and Shahar 2008) and can act as a healthy coping strategy for managing the stressors of running away and becoming homeless (Williams, Lindsey, Kurtz and Jarvis 2001). Mechanisms of resilience utilized by homeless young adults, may also extend to other homophobic, heteronormative environments, such as the college campus.

In stark contrast to the LGBTQ homeless young adults in this study, LGBTQ college students shared overwhelmingly positive accounts of support and acceptance on the college campus, which helped to develop their sense of resilience in response to identity-related challenges. Therefore, the campus environment acted as a positive complement to college students' resiliency, so that it bolstered their belief that their gender and sexual identity enhanced their lives. Other studies support this finding, especially for trans students who experienced positive affirmation on their college

campus through perceiving inclusive language, health care and student ally groups (Singh, Meng and Hansen 2013). Developing resiliency and conceptualizing identity-related challenges as life enhancement is especially important for LGBTQ college students because of the persistence of heteronormative campus climates (Wickens and Sandlin 2010) and the unique risks they face in college that can lead to dropping out of college, such as feelings of social isolation and lack of institutional support for LGBT issues (Mancini 2011).

The theme of "Identity as Life Enhancement" also has significant implications for understanding LGBTQ young adults' life course trajectories and their transitions through emerging adulthood. Though the accumulation of risk can create cumulative disadvantages across a person's life, such as undergoing homelessness during the period of adolescence or enduring family rejection, experiences that work to counteract adverse life course events can potentially buffer these negative effects (Elder 1998; Kennedy et al. 2010). Relatedly, developing resilience and other positive coping strategies in emerging adulthood, as having adaptive resources in response to conflicts and struggles, such as effective problem-solving and rational thinking, has been shown to predict the successful transition into adulthood (Masten et al. 2004). Encouraging and promoting resilience and personal competency in young people can have profound effects in orienting their life course onto a healthy path, rather than only focusing on challenges and risks (Schoon and Bynner 2003).

This study showed that young people's ability to develop resiliency in the face of adversity was highly contingent upon context and social environment, therefore it is necessary to consider how well a sense of resilience can successfully carry LGBTQ

young adults into middle and later life. While scholars highlight the ways that resiliency can buffer the negative effects of victimization for LGBT youth, contextual factors can diminish this relationship, such as in rural areas where LGBT-directed violence is higher (Kosciw, Palmer and Kull 2015). From a life course perspective, developing resilience can be a vital component to sustaining mental and physical health, as well as enduring pivotal, and possibly challenging, life course transitions, especially for sexual and gender minority young people (Saewyc 2011). For example, while homeless young adults were adamant in this study to conceptualize their struggles as positive, life-enhancing experiences, the question remains as to whether this heightened sense of resiliency can imbue them with tangible skills and opportunities that allow them to exit homelessness and improve their life conditions. Connections to broader, community-based LGBTQ support groups are especially crucial for young people in providing them with accessible resources throughout their life course (Allen, Hammack and Himes 2012) that are not contingent upon a particular social context, such as campus or homelessness.

## Implications for Service, Policy and Law

These findings have important implications for policymakers, service providers and lawmakers. On the college campus, educators and administrators must remain vigilant in making the classroom and wider campus context a welcoming, inclusive environment for LGBTQ young adults. Exploring the intricacies of the campus context is key in examining the nuances of LGBTQ college students' experiences, so that college officials can address more homophobic and heteronormative environments, such as fraternities (Worthen 2014). Furthermore, colleges and universities should work to implement more programs and resources for LGBT college students, as research has

identified the positive influence of programs, such as the LGBT Safe Zone, including increasing awareness of and support for LGBT-identified people and issues on campus (Evans 2002). Campus officials must also be aware of how LGBTQ college students' multiple, intersecting identities, such as sexuality, gender, race, class and religion, can create unique experiences for them and policies should be pursued that help to integrate students' identities and help them establish a sense of campus community (Poynter and Washington 2005). As college student participants in this study emphasized the importance of engaging in LGBTQ activism and advocacy on campus in developing their identities, college officials should continue to work to provide opportunities for LGBTQ leadership and advocacy that allow students to become involved and shape campus climate surrounding LGBTQ issues (Renn 2007).

Similarly, homeless service providers should take into account the unique challenges faced by LGBTQ homeless young adults in understanding how they tailor services and programs for these young people. For example, service providers should consider how agency ideology, particularly when it is religious-oriented, could create unwelcoming environments for these young people if they perceive prejudice and discrimination directed at their sexuality or gender. Homeless shelters are an example of one such site that can be perceived as hostile environments for LGBTQ young adults, especially if their rules dictate homophobic or transphobic regulations, such as prohibiting young people from sharing rooms/beds with their partners or not allowing trans people to use the bathroom reflective of their preferred gender (Mottet and Ohle 2006). Additionally, homeless youth shelters should be especially mindful of the disproportionate rates of victimization of LGBTQ runaway and homeless youth (Cochran

et al. 2002; Tyler 2008) and work to provide added protections and safeguards for these young people while they are in the shelter's care (Hunter 2008).

On a broader community level, lawmakers and policymakers in the United States and beyond must take steps to address the distinctive challenges that LGBTQ young adults encounter, as prejudice and discrimination experienced by LGBTQ college students and homeless young adults does not exist within a contextual vacuum. For example, broad reaching community education efforts aimed at LGBTQ issues, including hate crimes and microaggressions, can begin to bridge the gaps that exist between campus initiatives and homeless agency and service programs (Mayer, Mimiaga, VanDerwarker, Goldhammer and Bradford 2007). A more contextually comprehensive approach to improving the lives of LGBTQ young adults can more fully address their needs across multiple domains of emerging adulthood, including, among others, the complexity of family formation beyond a heteronormative model (Rabun and Oswald 2009) and the challenges they may experience in the workforce such as gender and sexuality-based discrimination (Willis 2012).

Structural changes to improve LGBTQ young adults' lives are required to address systemic failings that diminish or ignore the unique challenges these young people face and the impact of contextual factors. Though the majority of LGBTQ college students in this study cited the widespread support and access to resources for sexual minority individuals, the religious ideology of college campuses must also be taken into consideration. Many Christian colleges, for example, espouse alienating campus policies that on one hand are inclusive of LGBTQ identity and on the other are discriminatory against any type of non-heterosexual behavior, forcing LGBTQ students to live in fear of

negative academic repercussions, including expulsion (Wheeler 2016). Furthermore, federal regulations must work to expand protections to LGBTQ students, such as Title IX, which currently protects students from gender-based prejudice and discrimination in schools, but this law fails to prohibit harassment based on sexual attraction and identity (National Women's Law Center 2012). Structural support for LGBTQ college students on various types of campuses can encourage these young people's successful transitions into young adulthood and beyond.

In addition to structural changes within academic institutions to acknowledge the needs of LGBTQ college students, macro policies that provide support for gender and sexual minorities in the wider community can also benefit more at-risk young adults, including LGBTQ homeless youth. Health policy changes are required to be fully inclusive of a diverse range of gender and sexual identities that often face prejudice and discrimination (Mulé et al. 2009). Lack of adequate health care access and resources is especially salient for homeless LGBTQ young people; therefore public policy changes must address the nuances within this population such as unique family background, geographical region and variation across victimization and substance use (Keuroghlian, Shtasel and Bassuk 2014). Equitable access to quality healthcare is critical for LGBTQ young people as they emerge into adulthood and navigate life course transitions and turning points, including the pursuit of family formation and health maintenance (Hoffman, Freeman and Swann 2009).

#### LIMITATIONS

This study has limitations that require consideration. First, both samples of LGBTQ young adults were recruited using convenience-sampling methods. This

recruitment strategy created a constrained sampling frame that captured a particular subset of LGBTQ college students and homeless young adults. For example, some college students were recruited from LGBTQ-related campus email listservs and social justice-oriented classes and majors, which could have primed these participants for thinking about their gender and sexual identities in contexts that expressly encouraged this type of critical thinking. Further, homeless young adults were primarily recruited from service agencies, so that this sample does not represent the experiences of homeless youth who do not use services and therefore are possibly at greater risk. Many of these homeless young adults were also involved in service agency-sponsored LGBTQ support groups, which highlights their desire to engage with processes of identity development. A number of studies have documented the difficulty of capturing representative samples of hard to reach populations, including the homeless and LGBTQ people (Abrams 2010). Future research should attempt to sample both college and homeless LGBTQ young adults using more diverse sampling methods to capture a wider breadth of these individuals' experiences.

The disparate knowledge of LGBTQ language and issues across the subsamples may also have influenced the data and how young people responded to interview questions. This was evident in this study in terms of young people's identities, such that college students expressed more variation in identifying outside of the more normative convention of the LGBTQ spectrum, such as being asexual, pansexual (romantic attraction without regard to gender categories) or a demi-girl (defined by participant as identifying mostly as feminine while also feeling comfortable expressing masculine traits). Additionally, many homeless respondents admitted that they were not familiar

with the LGBTQ acronym and had more difficulty understanding explanations of each of these terms. Disparities in knowledge and education surrounding identity-related matters could shape how LGBTQ young adults understand their own sexuality and gender and engage with identity development on individual, interpersonal and community levels (DeHaan et al. 2013).

It is also necessary to acknowledge the barrier of gaining access to participants and how this may have influenced the data. For example, social desirability bias could have affected how respondents' disclosed information based on the sensitive nature of the interviews concerning their sexual and gender identities and the social stigma attached to these statuses. In particular, the methodological format of face-to-face interviewing, compared to surveys, may have led to respondents' greater hesitancy in sharing more personal anecdotes, such as experiences of victimization or sexual activity. Additionally, the insider/outsider dilemma posed challenges in gaining the trust of participants when a researcher does not share membership with the population of interest (Levy 2013). For example, some homeless youth expressed their reluctance to speak with me because they viewed me as an authority figure, and therefore they were likely to mistrust my purpose and intentions. Gaining the trust of various gatekeepers to homeless young adult networks, such as service providers and leaders of homeless peer groups, however, enhanced the legitimacy of my study and my role as a researcher, and this prompted some youth to participate in my study. On the other hand, my role as a graduate student whose gender and sexual identities were unknown to participants allowed me to build rapport with college students as we inhabited the shared role of student and could commiserate on similar stressors we faced in pursuing higher education. In these ways, embodying

both an insider and outsider perspective in varying aspects simultaneously acted as a barrier and a facilitator to recruitment and data collection.

Another barrier to this study is the retrospective nature of the young people's accounts that were captured at one point in time and its limited definition of how young adults can experience sexual and gender minority statuses. The young adults' memories of past experiences may have been flawed by lapses in memory, misremembering the chronology of events and the exaggeration or minimization of life experiences (Hardt and Rutter 2004). Furthermore, LGBTQ young adults may misremember their coming out experiences, in particular, especially if this process led to traumatic events that they have blocked from or minimized in their memory as a coping strategy (Rosario, Hunter, Maguen, Gwadz and Smith 2001). Relatedly, this study only addressed how sexual and gender identity shaped young people's lives, as it was an eligibility requirement that participants claim a particular sexual and/or gender minority status. Sexual attraction, rather than identity, however, may be more indicative of experiences of prejudice, discrimination and overall well-being because of its potential greater visibility and public nature for young people (Johns, Zimmerman and Bauermeister 2013).

The cross-sectional nature of the data itself being collected in a single interview also warrants discussion. Capturing the breadth of one's life course in one interview is challenging in that it forces a respondent to recall a vast amount of information, including their early lives and the sequencing of life events. While this study is useful in providing a snapshot of LGBTQ young adults' lives, qualitative longitudinal research could help ease these concerns by allowing for a wider scope of their life course and how certain experiences may act as pivotal turning points. Researchers must be aware, however, of

the difficulty in tracking more marginalized populations, such as homeless youth, and the increased risk of attrition in pursuing qualitative, longitudinal studies (Maycock and Corr 2013). Despite these drawbacks, one-time face-to-face interviews allowed the young adults to conceptualize their life course trajectories from their own perspectives, rather than a researcher interpreting the young people's lives from an outside point of view.

#### **FUTURE RESEARCH**

This study of LGBTQ young adults also points to numerous fruitful avenues for future research. For example, prospective studies would benefit from incorporating a participatory action research model to overcome some of the aforementioned limitations by enlisting LGBTQ college students and homeless young adults to act as researchers in collecting data from their peers. This method would allow participants an active stake in the research process and provide more opportunities for effecting real change in issues that marginalized groups are facing, such as health and social disparities among homeless populations (Yeich 1996) and LGBTQ individuals (Northridge, McGrath and Krueger 2007). For young people, involvement in participatory research could also act as a catalyst to their engagement with collective action and social movements in the wider society that enhances their sense of social responsibility (Fox et al. 2010).

Furthermore, future research should explore the experiences of LGBTQ young adults who simultaneously inhabit multiple, overlapping social contexts and how these environments shape their experiences. One such example of this would be young people who are both homeless and attending college, as research on this hard-to-reach population is nearly non-existent. College students in particular may be pushed to homelessness due to the inability to afford housing and other financial concerns related to

the growing costs of higher education (Ringer 2015). LGBTQ homeless college students, then, would encounter the dual stigma of residential instability and having non-conforming gender and sexual identities, highlighting the importance of examining how they navigate differing social contexts and environmental influences in their lives. Furthermore, wider regional context is also important to consider in its unique impact on LGBTQ young adults, such as how their experiences vary in more politically conservative states that legally allow discrimination based on sexual and gender orientation in rental housing (Lauster and Easterbrook 2011) and employment (Sears and Mallory 2011). A lack of legal protections can exacerbate the challenges faced by LGBTQ young people, particularly those who are homeless and/or in college.

By simultaneously exploring the life course experiences of LGBTQ emerging adults who are in college and those who are experiencing homelessness, this study qualitatively illuminates the distinctive ways that social context can shape young people's experiences and perspectives. While both LGBTQ homeless young adults and LGBTQ college students strongly emphasized the importance of sexual and gender identity within their lives, their primary social contexts of either the campus or homelessness impacted how they were able to engage with identity development. By acting as a source of identity-related privilege, social context worked to act as either a hindrance or a facilitating factor in how young adults could focus on their gender and sexuality. The influential role of social context oftentimes encouraged the LGBTQ young people to manage their identity in unique ways, which was also closely tied to their social relationships.

Strategies concerning how LGBTQ young adults managed their identities in various social contexts and relationships emerged as a formative aspect of their lives, particularly as they navigated pathways to adulthood. Understanding how LGBTQ young people of various social backgrounds and social contexts traverse identity-related challenges and develop effective coping strategies is essential in highlighting the diversity of this population. Furthermore, strategically managing one's LGBTQ identity can endure across the life course and shift over time, contexts and relationships, so these strategies may require continual assessment and modification. Though these young people experienced a number of different challenges in navigating and managing their identities, they did not always view these experiences as negative or debilitating, but rather often construed them as intrinsically rewarding.

The majority of research on LGBTQ young adults tends to emphasize the risks and dangers that they face as a result of cultural and social prejudice and discrimination. This study shows, however, the ways that LGBTQ young people, regardless of social context, can work to overcome these issues and develop resiliency in the face of adversity. Exploring how LGBTQ young adults conceptualize their identity as a life enhancement can broaden explanations for how they can successfully transition into adulthood and later life. Emphasizing resiliency in tandem with a risks-based framework can help scholars, service providers and policy makers in enhancing the lives of marginalized populations. Taken together, the findings from this study provide a more complete, nuanced picture of sexual and gender minority young adults through a life course exploration of their social contexts, relationships and identity development trajectories.

#### References

- Abes, Elisa S., and David Kasch. 2007. "Using Queer Theory to Explore Lesbian College Students' Multiple Dimensions of Identity." *Journal of College Student Development* 48:619-636.
- Abes, Elisa S., Susan R. Jones, and Marylu K. McEwen. 2007. "Reconceptualizing the Model of Multiple Dimensions of Identity: The Role of Meaning-Making Capacity in the Construction of Multiple Identities." *Journal of College Student Development* 48:1-22.
- Abes, Elisa S. and Susan R. Jones. 2004. "Meaning-Making Capacity and the Dynamics of Lesbian College Students' Multiple Dimensions of Identity." *Journal of College Student Development* 45:612-632.
- Abrams, Laura S. "Sampling 'Hard to Reach' Populations in Qualitative Research The Case of Incarcerated Youth." *Qualitative Social Work* 9:536-550.
- Allen, Kenneth D., Phillip L. Hammack and Heather L. Himes. 2012. "Analysis of GLBTQ Youth Community-Based Programs in the United States." *Journal of Homosexuality* 59:1289-1306.
- Angen, Maureen Jane. 2000. "Evaluating Interpretive Inquiry: Reviewing the Validity Debate and Opening the Dialogue." *Qualitative Health Research* 10:378-395.
- Arnett, Jeffrey Jensen. 2015. "College Students as Emerging Adults: The Developmental Implications of the College Context." *Emerging Adulthood*. DOI: 2167696815587422.

- Arnett, Jeffrey Jensen, Marion Kloep, Leo B. Hendry, and Jennifer L. Tanner. 2011.

  \*Debating Emerging Adulthood: Stage or Process? New York, NY: Oxford University Press.
- Arnett, Jeffrey Jensen. 2005. "The Vitality Criterion: A New Standard of Publication for Journal of Adolescent Research." *Journal of Adolescent Research* 20: 3-7.
- Arnett, Jeffrey Jensen. 2000. "Emerging Adulthood: A Theory of Development from the Late Teens Through the Twenties." *American Psychologist* 55:469-480.
- Auerswald, Colette L., and Stephen L. Eyre. 2002. "Youth Homelessness in San Francisco: A Life Cycle Approach." *Social Science & Medicine* 54:1497-1512.
- Berg, Bruce Lawrence and Howard Lune. 2004. *Qualitative Research Methods for the Social Sciences*. Boston, MA: Pearson.
- Bidell, Markus P. 2014. "Is There an Emotional Cost of Completing High School? Ecological Factors and Psychological Distress Among LGBT Homeless Youth." *Journal of Homosexuality* 61:366-381.
- Bilodeau, Brent L. and Kristen A. Renn. 2005. "Analysis of LGBT Identity Development Models and Implications for Practice." *New Directions for Student Services* 2005:25-39.
- Bouris, Alida, Vincent Guilamo-Ramos, Angela Pickard, Chengshi Shiu, Penny S.

  Loosier, Patricia Dittus, Kari Gloppen and J. Michael Waldmiller. 2010. "A

  Systematic Review of Parental influences on the Health and Well-Being of

  Lesbian, Gay, and Bisexual Youth: Time for a New Public Health Research and

  Practice Agenda." *The Journal of Primary Prevention* 31:273-309.

- Bowen, Glenn A. 2008. "Naturalistic Inquiry and the Saturation Concept: A Research Note." *Qualitative Research* 8:137-152.
- Boyatzis, Richard E. 1998. *Transforming Qualitative Information: Thematic Analysis and Code Development*. Thousand Oaks, CA: Sage Publications.
- Bregman, Hallie R., Neena M. Malik, Matthew J. L. Page, Emily Makynen and Kristin M. Lindahl. 2013. "Identity Profiles In Lesbian, Gay, And Bisexual Youth: The Role of Family Influences." *Journal of Youth and Adolescence* 42:417-430.
- Center for American Progress. 2010. "Gay and Transgender Youth Homelessness by the Numbers." Retrieved February 1, 2015 from https://www.americanprogress.org/issues/lgbt/news/2010/06/21/7980/gay-and-transgender-youth-homelessness-by-the-numbers/.
- Caspi, Avshalom and Brent W. Roberts. 2001. "Personality Development Across the Life Course: The Argument for Change and Continuity." *Psychological Inquiry* 12:49-66.
- Charmaz, Kathy. 2014. *Constructing Grounded Theory*. Thousand Oaks, CA: Sage Publications.
- Charmaz, Kathy and Belgrave, Linda Liska. 2002. "Qualitative Interviewing and Grounded Theory Analysis." Pp. 347-365 in *The SAGE Handbook of Interview Research: The Complexity of the Craft*, edited by J. F. Gubrium, J. A. Holstein, A. B. Marvasti & K. D. McKinney. Thousand Oaks, CA: Sage Publications.
- Chen, Xiaojin, Kimberly A. Tyler, Les B. Whitbeck and Dan R. Hoyt. 2004. "Early Sexual Abuse, Street Adversity, and Drug Use Among Female Homeless and Runaway Adolescents in the Midwest." *Journal of Drug Issues* 34:1-21.

- Clausen, John. A. 1991. "Adolescent Competence and the Shaping of the Life Course." *American Journal of Sociology* 96:805-842.
- Cochran, Bryan N., Angela J. Stewart, Joshua A. Ginzler and Ana Mari Cauce. 2002.

  "Challenges Faced by Homeless Sexual Minorities: Comparison of Gay, Lesbian,
  Bisexual, and Transgender Homeless Adolescents with their Heterosexual

  Counterparts." *American Journal of Public Health* 92:773-777.
- Collins, Patricia Hill. 2009. *Black Feminist Thought: Knowledge, Consciousness and the Politics of Empowerment*. New York: Routledge.
- Côté, James E. and Charles G. Levine. 2002. *Identity, Formation, Agency, And Culture:*A Social Psychological Synthesis. Mahwah, NJ: Lawrence Erlbaum Associates.
- Couch, Jen, Ben Durant and Jennifer Hill. 2012. "Young People, Old Issues:

  Methodological Concerns in Research with Highly Marginalised Young
  People" *Youth Studies Australia* 31:46-54.
- Craig, Shelley L., Edmon W. Tucker and Eric F. Wagner. 2008. "Empowering Lesbian, Gay, Bisexual, and Transgender Youth: Lessons Learned From a Safe Schools Summit." *Journal of Gay & Lesbian Social Services* 20:237-252.
- Creswell, John W. and Dana L. Miller. 2010. "Determining Validity in Qualitative Inquiry." *Theory into Practice* 39:124-130.
- Creswell, John W. 2013. *Qualitative Inquiry and Research Design: Choosing Among Five Approaches*. Thousand Oaks, CA: Sage Publications.
- D'Augelli, Anthony R. 2002. "Mental Health Problems Among Lesbian, Gay, and Bisexual Youths Ages 14 to 21." *Clinical Child Psychology and Psychiatry* 7:433-456.

- D'Augelli, Anthony R., Arnold H. Grossman and Michael T. Starks. 2005. "Parents'

  Awareness of Lesbian, gay, and Bisexual Youths' Sexual Orientation." *Journal of Marriage and Family* 67:474-482.
- D'Augelli, Anthony R., Arnold H. Grossman, Nicholas P. Salter, Joseph J. Vasey,

  Michael T. Starks and Katerina O. Sinclair. 2005. "Predicting the Suicide

  Attempts of Lesbian, Gay, and Bisexual Youth." Suicide and Life-Threatening

  Behavior 35:646-660.
- D'Augelli, Anthony R. 1993. "Preventing Mental Health Problems among Lesbian and Gay College Students." *Journal of Primary Prevention* 13:245-261.
- D'Augelli, Anthony R. 1992. "Lesbian and Gay Male Undergraduates' Experiences of Harassment and Fear on Campus." *Journal of Interpersonal Violence* 7:383-395.
- D'Augelli, Anthony R. 1991. "Gay Men in College: Identity Processes and Adaptations." *Journal of College Student Development* 32:140-146.
- DeHaan, Samantha, Laura E. Kuper, Joshua C. Magee, Lou Bigelow, and Brian S.

  Mustanski. 2013. "The Interplay Between Online and Offline Explorations Of Identity, Relationships, and Sex: A Mixed-Methods Study With LGBT Youth."

  Journal of Sex Research 50:421-434.
- Denes, Amanda and Tamara D. Afifi. 2014. "Coming Out Again: Exploring GLBQ Individuals' Communication with Their Parents After the First Coming Out."

  Journal of GLBT Family Studies 10:298-325.
- Diamond, Lisa M. and Molly Butterworth. 2008. "Questioning Gender And Sexual Identity: Dynamic Links Over Time." *Sex Roles* 59:365-376.

- Dickson-Swift, Virginia, Erica L. James, Sandra Kippen and Pranee Liamputtong. 2006. "Blurring Boundaries in Qualitative Health Research on Sensitive Topics."

  Qualitative Health Research 16:853-871.
- Doty, Nathan Daniel, Brian L. B. Willoughby, Kristin M. Lindahl and Neena M. Malik. 2010. "Sexuality Related Social Support Among Lesbian, Gay, and Bisexual Youth." *Journal of Youth and Adolescence* 39:1134-1147.
- Durso, Laura E. and Gary J. Gates. 2012. "Serving Our Youth: Findings from a National Survey of Service Providers Working with Lesbian, Gay, Bisexual, and Transgender Youth who are Homeless or At Risk of Becoming Homeless." Los Angeles, CA: The Williams Institute with True Colors Fund and The Palette Fund.
- Elder, Glen H. 1998. "The Life Course as Developmental Theory." *Child Development* 69:1-12.
- Elder, Glen H. 1994. "Time, Human Agency, and Social Change: Perspectives on the Life Course." *Social Psychology Quarterly* 57:4-15.
- Elder, Glen H., Jr., Monica Kirkpatrick Johnson, and Robert Crosnoe. 2003. "The Emergence and Development of Life Course Theory." Pp. 3-19 in *Handbook of the Life Course*, edited by J. T. Mortimer and M. J. Shanahan. New York: Kluwer Academic/Plenum Publishers.
- Evans, Nancy J. and Ellen M. Broido. 1999. "Coming Out in College Residence Halls:

  Negotiation, Meaning Making, Challenges, Supports." *Journal of College Student Development* 40:658-668.

- Evans, Nancy. 2002. "The Impact of an LGBT Safe Zone Project on Campus Climate." *Journal of College Student Development* 43:522-539.
- Evans, Nancy J. and Ellen M. Broido. 2002. "The Experiences of Lesbian and Bisexual Women in College Residence Halls: Implications for Addressing Homophobia and Heterosexism." *Journal of Lesbian Studies* 6:29-42.
- Fergus, Stevenson and Marc A. Zimmerman. 2005. "Adolescent Resilience: A Framework for Understanding Healthy Development in the Face of Risk." *Annual Review of Public Health* 26:399-419.
- Ferguson, Kristin M. 2009. "Exploring Family Environment Characteristics and Multiple Abuse Experiences Among Homeless Youth." *Journal of Interpersonal Violence* 24:1875-1891.
- Finkelstein, Marni. 2005. With No Direction Home: Homeless Youth on the Road and in the Streets. Belmont, CA: Thomson Wadsworth.
- Fox, Madeline, Kavitha Mediratta, Jessica Ruglis, Brett Stoudt, Seema Shah, and Michelle Fine. 2010. "Critical Youth Engagement: Participatory Action Research and Organizing." Pp. 621-650 in *Handbook Of Research On Civic Engagement In Youth*, edited by L. R. Sherrod, J. Torney-Purta & C. A. Flanagan. Hoboken, NJ: John Wiley & Sons, Inc.
- Franzosi, Roberto. 1998. "Narrative Analysis-or Why (and How) Sociologists Should be Interested in Narrative." *Annual Review of Sociology* 24:517-554.
- Gangamma, Rashmi, Natasha Slesnick, Paula Toviessi and Julianne Serovich. 2008. "Comparison of HIV Risks Among Gay, Lesbian, Bisexual and Heterosexual Homeless Youth." *Journal of Youth and Adolescence* 37:456-464.

- Gates, Gary J. and Frank Newport. 2012. "Special Report: 3.4% of U.S. Adults Identify as LGBT." Gallup. Retrieved January 22, 2015 from http://www.gallup.com/poll/158066/special-report-adults-identify-lgbt.aspx.
- Gattis, Maurice N. 2013. "An Ecological Systems Comparison Between Homeless Sexual Minority Youths and Homeless Heterosexual Youths." *Journal of Social Service Research* 39:38-49.
- Gattis, Maurice N. and Andrea Larson. 2016. "Perceived Racial, Sexual Identity, and Homeless Status-Related Discrimination among Black Adolescents and Young Adults Experiencing Homelessness: Relations with Depressive Symptoms and Suicidality." *American Journal of Orthopsychiatry* 86:79-90.
- George, Linda K. 1993. "Sociological Perspectives on Life Transitions." *Annual Review of Sociology* 19:353-373.
- Glassgold, Judith M. 2007. "In Dreams Begin Responsibilities': Psychology, Agency, and Activism." *Journal of Gay & Lesbian Psychotherapy* 11:37-57.
- Goldfried, Marvin R., and Anita P. Goldfried. 2001. "The Importance of Parental Support in the Lives of Gay, Lesbian, and Bisexual individuals." *Journal of Clinical Psychology* 57:681-693.
- Guest, Greg, Arwen Bunce and Laura Johnson. 2006. "How Many Interviews Are Enough? An Experiment with Data Saturation and Variability." *Field Methods* 18:59-82.
- Hackimer, Laura and Sherrie L. Proctor. 2015. "Considering the Community Influence for Lesbian, Gay, Bisexual, and Transgender Youth." *Journal of Youth Studies* 8:277-290.

- Hardt, Jochen and Michael Rutter. 2004. "Validity of Adult Retrospective Reports of Adverse Childhood Experiences: Review of the Evidence." *Journal of Child Psychology and Psychiatry* 25:260-273.
- Hoffman, Neal D., Katherine Freeman and Stephanie Swann. 2009. "Healthcare Preferences of Lesbian, Gay, Bisexual, Transgender and Questioning Youth."

  \*\*Journal of Adolescent Health 45:222-229.\*\*
- Hoffman, Saul D., E. Michael Foster and Frank F. Furstenberg Jr. 1993. "Reevaluating the Costs of Teenage Childbearing." *Demography* 30:1-13.
- Howard, Adam and Rubén A. Gaztambide-Fernandez. 2010. *Educating Elites: Class Privilege and Educational Advantage*. Lanham, MD: Rowman & Littlefield Education.
- Hunter, Ernst. 2008. "What's Good for the Gays is Good for the Gander: Making Homeless Youth Housing Safer for Lesbian, Gay, Bisexual, and Transgender Youth." *Family Court Review* 46:543-557.
- Jackson, Stevi. 2006. "Gender, Sexuality and Heterosexuality: The Complexity (and Limits) of Heteronormativity." *Feminist Theory* 7:105-121.
- Johns, Michelle Marie, Marc Zimmerman and Jose A. Bauermeister. 2013. "Sexual Attraction, Sexual Identity, and Psychosocial Wellbeing in a National Sample of Young Women during Emerging Adulthood." *Journal of Youth and Adolescence* 24:82-95.
- Jones, Susan R. and Marylu K. McEwen. 2000. "A Conceptual Model of Multiple

  Dimensions Of Identity." *Journal of College Student Development* 41:405-414.

- Kennedy, Angie C., DeBrenna LaFa Agbenyiga, Natalie Kasiborski and Jessica Gladden.
  2010. "Risk Chains over the Life Course Among Homeless Urban Adolescent
  Mothers: Altering Their Trajectories Through Formal Support." Children and
  Youth Services Review 32:1740-1749.
- Kerr, Dianne L., Kele Ding and Julie Chaya. 2014. "Substance Use of Lesbian, Gay, Bisexual and Heterosexual College Students." *American Journal of Health Behavior* 38:951-962.
- Keuroghlian, Alex S., Derri Shtasel and Ellen L. Bassuk. 2014. "Out on the Street: A Public Health and Policy Agenda for Lesbian, Gay, Bisexual, and Transgender Youth Who Are Homeless." *American Journal of Orthopsychiatry* 84:66-72.
- Kidd, Sean and Golan Shahar. 2008. "Resilience in Homeless Youth: The Key Role of Self-Esteem." *American Journal of Orthopsychiatry* 78:163-172.
- Koltko-Rivera, Mark E. 2006. "Rediscovering the Later Version of Maslow's Hierarchy of Needs: Self-Transcendence and Opportunities for Theory, Research, and Unification." *Review of General Psychology* 10:302-317.
- Kosciw, Joseph G., Neal A. Palmer and Ryan M. Kull. 2015. "Reflecting Resiliency:

  Openness About Sexual Orientation and/or Gender Identity and Its Relationship

  to Well-Being and Educational Outcomes for LGBT Students." *American Journal*of Community Psychology 55:167-178.
- LaSala. Michael C. 2010. Coming Out, Coming Home: Helping Families Adjust to a Gay or Lesbian Child. New York, NY: Columbia University.

- Lauster, Nathanael and Adam Easterbrook. 2011. "No Room For New Families? A Field Experiment Measuring Rental Discrimination Against Same-Sex Couples and Single Parents." *Social Problems* 58:389-409.
- Lee, Yeon-Ok, and Raymond M. Lee. 2012. "Methodological Research on "Sensitive"

  Topics: A Decade Review." *Bulletin of Sociological Methodology/Bulletin de Méthodologie Sociologique* 114:35-49.
- Lee, Barrett A., Kimberly A. Tyler and James D. Wright. 2010. "The New Homelessness Revisited." *Annual Review of Sociology* 36:501-521.
- Levy, Denise L. 2013. "On the Outside Looking In? The Experience of Being a Straight,

  Cisgender Qualitative Researcher." *Journal of Gay & Lesbian Social*Services 25:197-209.
- Loh, Jason. 2013. "Inquiry into Issues of Trustworthiness and Quality in Narrative Studies: A Perspective." *The Qualitative Report* 18:1-15.
- Longerbeam, Susan D., Karen Kurotsuchi Inkelas, Dawn R. Johnson and Zakiya S. Lee. 2007. "Lesbian, Gay, and Bisexual College Student Experiences: An Exploratory Study." *Journal of College Student Development* 48:215-230.
- Love, Patrick, Marianne Bock, Annie Jannarone and Paul Richardson. 2005. "Identity Interaction: Exploring the Spiritual Experiences of Lesbian and Gay College Students." *Journal of College Student Development* 46:193-209.
- Maccio, Elaine M. and Kristin M. Ferguson. 2016. "Services to LGBTQ Runaway and Homeless Youth: Gaps and Recommendations." *Children and Youth Services Review* 63:47-57.

- Mallett, Shelley and Doreen Rosenthal. 2009. "Physically Violent Mothers Are a Reason for Young People's Leaving Home." *Journal of Interpersonal Violence* 24:1165-1174.
- Mallett, Shelley, Doreen Rosenthal and Deborah Keys. 2005. "Young People, Drug Use and Family Conflict: Pathways into Homelessness." *Journal of Adolescence* 28:185-199.
- Mancini, Olivia. 2011. "Attrition Risk and Resilience Among Sexual Minority College Students." *Columbia Social Work Review* 2:8-22.
- Marshal, Michael P., Laura J. Dietz, Mark S. Friedman, Ron Stall, Helen Smith, James McGinley, Brian C. Thoma, Pamela J. Murray, Anthony R. D'Augelli, and David A. Brent. 2011. "Suicidality and Depression Disparities Between Sexual Minority And Heterosexual Youth: A Meta-Analytic Review." *Journal of Adolescent Health* 49:115-123.
- Masten, Ann S., Keith B. Burt, Glenn I. Roisman, Jelena Obradovic, Jeffrey D. Long and Auke Tellegen. 2004. "Resources and Resilience in the Transition to Adulthood:

  Continuity and Change." *Development and Psychopathology* 16:1071-1094.
- Maycock, Paula, and Mary-Louise Corr. 2013. "Young People's Homeless And Housing Pathways: Key Findings From A 6-Year Qualitative Longitudinal Study." Dublin, Ireland: Department of Children and Youth Affairs, Government Publications.
- Mayer, Kenneth H., Matthew J. Mimiaga, Rodney VanDerwarker, Hilary Goldhammer and Judith B. Bradford. 2007. Pp. 693-715 in *The Health of Sexual Minorities*, edited by I. H. Meyer & M. E. Northridge. New York: Springer.

- McCabe, Paul C., Eliza A. Dragowski and Florence Rubinson. 2013. "What is

  Homophobic Bias Anyway? Defining and Recognizing Microaggressions and

  Harassment of LGBTQ Youth." *Journal of School Violence* 12:7-26.
- McDermott, Elizabeth, Katrina Roen and Jonathan Scourfield. 2008. "Avoiding Shame: Young LGBT People, Homophobia and Self-Destructive Behaviours." *Culture, Health & Sexuality* 10:815-829.
- McLeod, Jane D. and Elbert P. Almazan. 2003. "Connections between Childhood and Adulthood." Pp. 391-412 in *The Handbook of the Life Course*, edited by J. T. Mortimer and M. J. Shanahan. New York: Kluwer Academic/Plenum Publishers.
- Merriam, Sharan B. and Elizabeth J. Tisdell. 2016. *Qualitative Research: A Guide to Design and Implementation*. San Francisco: John Wiley & Sons, Inc.
- Meyer, Ilan H. 2003. "Prejudice, Social Stress, and Mental Health in Lesbian, Gay, and Bisexual Populations: Conceptual Issues and Research Evidence." *Psychological Bulletin* 129:674-697.
- Milburn, Norweeta G., Francisco Javier Iribarren, Eric Ric, Marguerita Lightfoot, Rosa Solorio, Mary Jane Rotheram-Borus, Katherine Desmond, Alex Lee, Kwame Alexander, Katherine Maresca, Karen Eastmen, Elizabeth Mayfield Arnold and Naihua Duan. 2012. "A Family Intervention to Reduce Sexual Risk Behavior, Substance Use, and Delinquency Among Newly Homeless Youth." *Journal of Adolescent Health* 50:358-364.
- Moen, Phyllis, and Elaine Hernandez. 2009. "Social Convoys: Studying Lives in Time, Context, and Motion." Pp. 258-279 in *The Craft of Life Course Research*, edited by Glen H. Jr. Elder and Janet Z. Giele. New York: The Guilford Press.

- Mottet, Lisa and John Ohle. 2006. "Transitioning our Shelters: Making Homeless Shelters Safe for Transgender People." *Journal of Poverty* 10:77-101.
- Mulé, Nick J., Lori E. Ross, Barry Deeprose, Beth E. Jackson, Andrea Daley, Anna
  Travers and Dick Moore. 2009. "Promoting LGBT Health and Wellbeing
  Through Inclusive Policy Development." *International Journal for Equity in Health* 8:1-11.
- Nadal, Kevin L., Yinglee Wong, Katie E. Griffin, Kristin Davidoff, and Julie Sriken.

  2014. "The Adverse Impact Of Racial Microaggressions On College Students'

  Self-Esteem." *Journal of College Student Development* 55:461-474.
- Nadal, Kevin L., Avy Skolnik and Yinglee Wong. 2012. "Interpersonal and Systemic Microaggressions Toward Transgender People: Implications for Counseling." *Journal of LGBT Issues in Counseling* 6:55-82.
- Nadal, Kevin L., Marie-Anne Issa, Jayleen Leon, Vanessa Meterko, Michelle Wideman and Yinglee Wong. 2011. "Sexual Orientation Microaggressions: 'Death by a Thousand Cuts' for Lesbian, Gay, and Bisexual Youth." *Journal of LGBT Youth* 8:234-259.
- National Center on Family Homelessness. 2011. "The Characteristics and Needs of Families Experiencing Homelessness." Needham, MA: National Center on Family Homelessness. Retrieved March 22, 2012 from <a href="https://www.familyhomelessness.org/media/147.pdf">www.familyhomelessness.org/media/147.pdf</a>.
- National Women's Law Center. 2012. "Title IX Protections from Bullying and Harassment in Schools: FAQs for LGBT or Gender Nonconforming Students and

- Their Families." Retrieved March 21, 2016 from https://www.nwlc.org/sites/default/files/pdfs/lgbt bullying title ix fact sheet.pdf
- Needham, Belinda L. and Erika L. Austin. 2010. "Sexual Orientation, Parental Support, and Health During the Transition to Young Adulthood." *Journal of Youth and Adolescence* 39:1189-1198.
- Nesmith, Andrea A., David L. Burton, and T. J. Cosgrove. 1999. "Gay, Lesbian, and Bisexual Youth and Young adults: Social Support in Their Own Words." *Journal of Homosexuality* 37:95-108.
- Newman, Bernie S. and Peter G. Muzzonigro. 1993. "The Effects Of Traditional Family Values On The Coming Out Process Of Gay Male Adolescents." *Adolescence* 28:213-226.
- Noell, John W. and Linda M. Ochs. 2001. "Relationship of Sexual Orientation to Substance Use, Suicidal Ideation, Suicide Attempts, and Other Factors in a Population of Homeless Adolescents." *Journal of Adolescent Health* 29:31-36.
- Northridge, Mary E., Brian P. McGrath and Sam Quan Krueger. 2007. "Using Community-Based Participatory Research To Understand and Eliminate Social Disparities In Health For Lesbian, Gay, Bisexual, and Transgender Populations." Pp. 455-470 in *The Health of Sexual Minorities*, edited by I. H. Meyer & M. E. Northridge. New York: Springer.
- O'Rand, Angela M. 2006. "Stratification and the Life Course: Life Course Capital, Life Course Risks, and Social Inequality." Pp. 145-162 in *Handbook of Aging and the Social Sciences* (6th Edition), edited by R. H. Binstock and L. K. George.

  Burlington, MA: Academic Press.

- Oliver, Daniel G., Julianne M. Serovich and Tina L. Mason. 2005. "Constraints and Opportunities with Interview Transcription: Towards Reflection in Qualitative Research." *Social Forces* 84:1273-1289.
- Padgett, Deborah K. 2008. *Qualitative Methods in Social Work Research*. Thousand Oaks, CA: Sage Publications.
- Pearson, Jennifer and Lindsey Wilkinson. 2013. "Family Relationships and Adolescent Well-Being: Are Families Equally Protective for Same-Sex Attracted Youth?" *Journal of Youth and Adolescence* 42:376-393.
- Perry Jr, William G. 1999. Forms of Intellectual and Ethical Development in the College Years: A Scheme. San Francisco, CA: Jossey-Bass Publishers.
- Pinto, Rogério M., Rita M. Melendez and Anya Y. Spector. 2008. "Male-to-Female Transgender Individuals Building Social Support and Capital From Within a Gender-Focused Network." *Journal of Gay & Lesbian Social Services* 20:203-220.
- Plotnick, Robert D. 2007. "Adolescent Expectations and Desires About Marriage and Parenthood." *Journal of Adolescence* 30:943-963.
- Poteat, V. Paul, Jerel P. Calz, and Hirokazu Yoshikawa. 2016. "Promoting Youth Agency Through Dimensions of Gay–Straight Alliance Involvement and Conditions that Maximize Associations." *Journal of Youth and Adolescence* DOI: 10.1007/s10964-016-0421-6.
- Poynter, Kerry John and Jamie Washington. 2005. "Multiple Identities: Creating

  Community on Campus for LGBT Students." New Directions for Student Services

  111:41-47.

- Rabun, Carl and Ramona Faith Oswald. 2009. "Upholding and Expanding the Normal Family: Future Fatherhood through the Eyes of Gay Male Emerging Adults." *Fathering* 7:269-285.
- Ragins, Belle Rose and John M. Cornwell. 2001. "Pink Triangles: Antecedents and Consequences of Perceived Workplace Discrimination Against Gay and Lesbian Employees." *Journal of Applied Psychology* 86:1244-1261.
- Rankin, Susan R. 2005. "Campus Climates for Sexual Minorities." *New Directions for Student Services* 2005:17-23.
- Rankin, Susan R. 2003. *Campus Climate for LGBT People: A National Perspective*. New York: National Gay and Lesbian Task Force Policy Institute.
- Rankin, Sue, Genevieve N. Weber, Warren J. Blumenfeld and Somjen Frazer. 2010. 2010

  State of Higher Education for Lesbian, Gay, Bisexual & Transgender People.

  Charlotte, NC: Campus Pride.
- Ream, Geofrey L. and Nicholas R. Forge. 2014. "Homeless Lesbian, Gay, Bisexual, and Transgender (LGBT) Youth in New York City: Insights from the Field." *Child Welfare* 93:7-22.
- Renn, Kristen A. 2007. "LGBT Student Leaders and Queer Activists: Identities of Lesbian, Gay, Bisexual, Transgender, and Queer Identified College Student Leaders and Activists." *Journal of College Student Development* 48:311-330.
- Rew, Lynn, Tiffany A. Whittaker, Margaret A. Taylor-Seehafer and Lorie R. Smith.

  2005. "Sexual Health Risks and Protective Resources in Gay, Lesbian, Bisexual, and Heterosexual Homeless Youth." *Journal for Specialists in Pediatric Nursing* 10:11-19.

- Rew, Lynn, Margaret Taylor-Seehafer, Margaret, Nancy Y. Thomas and Ronald D.

  Yockey. 2001. "Correlates of Resilience in Homeless Adolescents." *Journal of Nursing Scholarship* 33:33-40.
- Rhoads, Robert A. 1994. *Coming Out in College: The Struggle for a Queer Identity*. Westport, CT: Greenwood Publishing.
- Riessman, Catherine Kohler. 2008. *Narrative Methods for the Human Sciences*.

  Thousand Oaks, CA: Sage Publications.
- Ringer, Brialle. 2015. "College Students Experiencing Homelessness: The Consequence of Failed Macro Policies." *McNair Scholars Research Journal* 8:103-124.
- Rosario, Margaret, Eric W. Schrimshaw and Joyce Hunter. 2012. "Homelessness Among Lesbian, Gay, and Bisexual Youth: Implications for Subsequent Internalizing and Externalizing Symptoms." *Journal of Youth and Adolescence* 41:544-560.
- Rosario, Margaret, Joyce Hunter, Shira Maguen, Marya Gwadz and Raymond Smith.

  2001. "The Coming-Out Process and Its Adaptational and Health-Related

  Associations among Gay, Lesbian, and Bisexual Youths: Stipulation and

  Exploration of a Model." *American Journal of Community Psychology* 29:133
  160.
- Russell, Stephen T., Caitlin Ryan, Russell B. Toomey, Rafael M. Diaz, and Jorge Sanchez. 2011. "Lesbian, Gay, Bisexual, and Transgender Adolescent School Victimization: Implications for Young Adult Health and Adjustment." *Journal of School Health* 81:223-230.
- Russell, Stephen T. 2005. "Beyond Risk: Resilience In The Lives Of Sexual Minority Youth." *Journal of Gay & Lesbian Issues in Education* 2:5-18.

- Ryan, Caitlin, Stephen T. Russell, David Huebner, Rafael Diaz and Jorge Sanchez. 2010. "Family Acceptance in Adolescence and the Health of LGBT Young Adults." *Journal of Child and Adolescent Psychiatric Nursing* 23:205-213.
- Ryan, Caitlin, David Huebner, Rafael M. Diaz and Jorge Sanchez. 2009. "Family Rejection as a Predictor of Negative Health Outcomes in White and Latino Lesbian, Gay, and Bisexual Young Adults." *Pediatrics* 123:346-352.
- Saewyc, Elizabeth M. 2011. "Research on Adolescent Sexual Orientation: Development,

  Health Disparities, Stigma, and Resilience." *Journal of Research on Adolescence* 21:256-272.
- Sanlo, Ronni. 2004. "Lesbian, Gay, and Bisexual College Students: Risk, Resiliency, and Retention." *Journal of College Student Retention: Research, Theory and Practice* 6:97-110.
- Savin-Williams, Ritch C. 2011. "Identity Development Among Sexual-Minority Youth."

  Pp. 671-689 in *Handbook of Identity Theory and Research*, edited by S. J.

  Schwartz, K. Luyckx and V. L. Vignoles. New York: Springer.
- Savin-Williams, Ritch C. and Geoffrey L. Ream. 2007. "Prevalence and Stability of Sexual Orientation Components During Adolescence and Young Adulthood." *Archives of Sexual Behavior* 36:385-394.
- Schoon, Ingrid and John Bynner. 2003. "Risk and Resilience in the Life Course:

  Implications for Interventions and Social Policies." *Journal of Youth Studies* 6:21-31.

- Schwartz, Seth J., James E. Côté and Jeffrey Jensen Arnett. 2005. "Identity And Agency In Emerging Adulthood: Two Developmental Routes In The Individualization Process." *Youth & Society* 37:201-229.
- Sears, Brad and Christy Mallory. 2011. "Documented Evidence of Employment

  Discrimination and Its Effects on LGBT People." The Williams Institute.

  Retrieved February 15, 2016 from http://eprints.cdlib.org/uc/item/03m1g5sg.
- Seil, Kacie S., Mayur M. Desai and Megan V. Smith. 2014. "Sexual Orientation, Adult Connectedness, Substance Use, and Mental Health Outcomes Among Adolescents: Findings From the 2009 New York City Youth Risk Behavior Survey." *American Journal of Public Health* 104:1950-1956.
- Shanahan, Michael J. 2000. "Pathways to Adulthood in Changing Societies: Variability and Mechanisms in Life Course Perspective." *Annual Review of Sociology* 26:667-692.
- Sheets, Raymond L. and Jonathan J. Mohr. 2009. "Perceived Social Support from Friends and Family and Psychosocial Functioning in Bisexual Young Adult College Students." *Journal of Counseling Psychology* 56:152-163.
- Shilo, Guy and Riki Savaya. 2011. "Effects of Family and Friend Support on LGB Youths' Mental Health and Sexual Orientation Milestones." *Family Relations* 60:318-330.
- Singh, Anneliese, Sarah Meng and Anthony Hansen. 2013. "'It's Already Hard Enough Being a Student': Developing Affirming College Environments for Trans Youth."

  \*\*Journal of LGBT Youth 10:208-223.\*\*

- Spencer, S. Melinda and Julie Hicks Patrick. 2009. "Social Support and Personal Mastery as Protective Resources During Emerging Adulthood." *Journal of Adult Development* 16:191-198.
- Stevens, Richard Allen. 2004. "Understanding Gay Identity Development within the College Environment." *Journal of College Student Development* 45:185-206.
- Tanner, Jennifer Lynn. 2006. "Recentering During Emerging Adulthood: A Critical
  Turning Point in Life Span Human Development." Pp. 21-55 in *Emerging Adults*in America: Coming of Age in the 21st Century, edited by J. J. Arnett and J. L.
  Tanner. Washington, D. C.: American Psychological Association.
- Thrane, Lisa E., Danny R. Hoyt, Les B. Whitbeck and Kevin A. Yoder. 2006. "Impact of Family Abuse on Running Away, Deviance and Street Victimization Among Homeless Rural and Urban Youth." *Child Abuse & Neglect* 30:1117-1128.
- Thompson, Sanna J., Kimberly Bender, Liliane Windsor, Mary S. Cook and Travonne Williams. 2010. "Homeless Youth: Characteristics, Contributing Factors and Service Options." *Journal of Human Behavior in the Social Environment* 20:193-217.
- Toomey, Russell B., Caitlin Ryan, Rafael M. Diaz, Noel A. Card, and Stephen T. Russell. 2010. "Gender-Nonconforming Lesbian, Gay, Bisexual, and Transgender Youth: School Victimization and Young Adult Psychosocial Adjustment."

  Developmental Psychology 46:1580-1589.
- Torkelson, Jason. 2012. "A Queer Vision of Emerging Adulthood: Seeing Sexuality in the Transition to Adulthood." *Sexuality Research and Social Policy* 9:132-142.

- Tyler, Kimberly A. 2008. "A Comparison of Risk Factors for Sexual Victimization Among Gay, Lesbian, Bisexual, and Heterosexual Homeless Young Adults." *Violence and Victims* 23:586-602.
- Tyler, Kimberly A. 2006. "A Qualitative Study of Early Family Histories and Transitions of Homeless Youth." *Journal of Interpersonal Violence* 21:1385-1393.
- Tyler, Kimberly A. and Lisa A. Melander. 2010. "Foster Care Placement, Poor Parenting, and Negative Outcomes Among Homeless Young Adults." *Journal of Child & Family Studies* 19:787-794.
- Tyler, Kimberly A. and Bianca E. Bersani. 2008. "A Longitudinal Study of Early

  Adolescent Precursors to Running Away." *Journal of Early Adolescence* 28:230251.
- Tyler, Kimberly A. and Les B. Whitbeck. 2004. Lost Childhoods: Risk and Resiliency Among Runaway and Homeless Adolescents. Pp. 378-397 in *Intervention with Children & Adolescents: An Interdisciplinary Perspective*, edited by P. Allen-Meares & M. W. Fraser. Boston.
- Tyler, Kimberly A. and Ana Mari Cauce. 2002. "Perpetrators of Early Physical and Sexual Abuse Among Homeless and Runaway Adolescents." *Child Abuse & Neglect* 26:1261-1274.
- Van Den Bergh, Nan and Catherine Crisp. 2004. "Defining Culturally Competent Practice with Sexual Minorities: Implications for Social Work Education and Practice." *Journal of Social Work Education* 40:221-238.

- Valentine, Gill, Tracey Skelton, and Ruth Butler. 2003. "Coming Out and Outcomes:

  Negotiating Lesbian And Gay Identities With, And In, The Family." *Environment and Planning D* 21:479-500.
- Van Leeuwen, James M., Susan Boyle, Stacy Salomonsen-Sautel, D. Nico Baker, J. T. Garcia, Allison Hoffman and Christian J. Hopfer. 2006. "Lesbian, Gay, and Bisexual Homeless Youth: An Eight-City Public Health Perspective." Child Welfare 85:151-170.
- Waller, Martha W. and Rebecca P. Sanchez. 2011. "The Association Between Same-Sex Romantic Attractions and Relationships and Running Away Among a Nationally Representative Sample of Adolescents." *Child and Adolescent Social Work Journal* 28:475-493.
- Westefeld, John S., Michael R. Maples, Brian Buford, and Steve Taylor. 2001. "Gay,

  Lesbian, and Bisexual College Students: The Relationship Between Sexual

  Orientation and Depression, Loneliness, and Suicide." *Journal of College Student*Psychotherapy 15:71-82.
- Wheeler, David R. 2016. "The LGBT Politics of Christian Colleges." The Atlantic.

  Retrieved March 21, 2016 from

  <a href="http://www.theatlantic.com/education/archive/2016/03/the-lgbt-politics-of-christian-colleges/473373/">http://www.theatlantic.com/education/archive/2016/03/the-lgbt-politics-of-christian-colleges/473373/</a>.
- Whitbeck, Les B., Xiaojin Chen, Dan R. Hoyt, Kimberly A. Tyler and Kurt D. Johnson. 2004. "Mental Disorder, Subsistence Strategies, and Victimization Among Gay, Lesbian, and Bisexual Homeless and Runaway Adolescents." *Journal of Sex Research* 41:329-342.

- Whitbeck, Les B. and Danny R. Hoyt. 1999. *Nowhere to Grow: Homeless and Runaway Adolescents and Their Families*. New York, NY: Walter de Gruyter.
- Whitbeck, Les B., Danny R. Hoyt and Shirley M. Huck. 1994. "Early Family Relationships, Intergenerational Solidarity, and Support Provided to Parents by Their Adult Children." *Journal of Gerontology* 49:S85-S94.
- Wickens, Corrine M., and Jennifer A. Sandlin. 2010. "Homophobia and Heterosexism in a College of Education: A Culture of Fear, a Culture of Silence." *International Journal of Qualitative Studies in Education* 23:651-670.
- Williams, Nancy R., Elizabeth W. Lindsey, P. David Kurtz and Sara Jarvis. 2001. "From Trauma to Resiliency: Lessons from Former Runaway and Homeless Youth." *Journal of Youth Studies* 4:233-253.
- Willis, Paul. 2012. "Witness on the Periphery: Young Lesbian, Gay, Bisexual and Queer Employees Witnessing Homophobic Exchanges in Australian Workplaces."

  Human Relations 65:1589-1610.
- Woodford, Michael R., Yoonsun Han, Shelley Craig, Colin Lim and Malinda M. Matney.

  2014. "Discrimination and Mental Health among Sexual Minority College

  Students: The Type and Form of Discrimination Does Matter." *Journal of Gay & Lesbian Mental Health* 18:142-163.
- Woodford, Michael R., Alex Kulick, Brandy R. Sinco and Jun Sung Hong. 2014.

  "Contemporary Heterosexism on Campus and Psychological Distress Among

  LGBQ Students: The Mediating Role of Self-Acceptance." *American Journal of Orthopsychiatry* 84:519-529.

- Woodford, Michael R., Michael L. Howell, Perry Silverschanz and Lotus Yu. 2012.

  "'That's So Gay!': Examining the Covariates of Hearing This Expression Among
  Gay, Lesbian, and Bisexual College Students." *Journal of American College*Health 60:429-434.
- Woronoff, Rob, Rudy Estrada, Susan Sommer, Michelle A. Marzullo, Caitlin Ryan, A.

  Chris Downs, H. J. David Ambroz and Peter Karys. 2006. *Out of the Margins: A Report on Regional Listening Forums Highlighting the Experiences of Lesbian, Gay, Bisexual, Transgender, and Questioning Youth in Care*. Washington, D. C.: Child Welfare League of America.
- Worthen, Meredith G.F. 2014. "Blaming the Jocks and the Greeks? Exploring Collegiate

  Athletes' and Fraternity/Sorority Members' Attitudes Toward LGBT

  Individuals." *Journal of College Student Development* 55:168-195.
- Wright, Eric R. and Brea L. Perry. 2006. "Sexual Identity Distress, Social Support, and the Health of Gay, Lesbian, and Bisexual Youth." *Journal of Homosexuality* 51:81-110.
- Yeich, Susan. 1996. "Grassroots Organizing with Homeless People: A Participatory Research Approach." *Journal of Social Issues* 52:111-121.
- Yoder, Kevin A., Dan R. Hoyt and Les B. Whitbeck. 1998. "Suicidal Behavior Among Homeless and Runaway Adolescents." *Journal of Youth and Adolescence* 27:753-771.

# **APPENDICES**

# Appendix A

7/10/2015

https://nugrant.unl.edu/era/orp/irb/viewPrintedMessage.php?ID=443291



May 28, 2015

Rachel Schmitz Department of Sociology

Kimberly Tyler Department of Sociology 717 OLDH, UNL, 68588-0324

IRB Number: 20150515214FB

Project ID: 15214

Project Title: On the Street and On Campus: A Comparison of Life Course Trajectories among Homeless

and College Lesbian, Gay, Bisexual and Transgender Young Adults

#### Dear Rachel:

This letter is to officially notify you of the approval of your project by the Institutional Review Board (IRB) for the Protection of Human Subjects. It is the Board's opinion that you have provided adequate safeguards for the rights and welfare of the participants in this study based on the information provided. Your proposal is in compliance with this institution's Federal Wide Assurance 00002258 and the DHHS Regulations for the Protection of Human Subjects (45 CFR 46).

You are authorized to implement this study as of the Date of Final Approval: 05/18/2015. This approval is Valid Until: 05/17/2016.

NOTE: Recruitment at UNO/UNMC may not take place until facilitated review is complete.

- \* Date of Approval: 05/18/2015
- \* Date of Full Board review/meeting: 04/29/2015
- Date of Chair review/acceptance of revisions: 05/18/2015
- \*Risk Classification: Minimal Risk
- \* Form Review: Expedited upon continuing and informational/minor changes (Expedited category 9)
- \* Funding: N/A
- Waiver Approval: Consent documentation under 45 CFR 46.116(d)(1-4)
- Subpart Review: N/A

We wish to remind you that the principal investigator is responsible for reporting to this Board any of the following events within 48 hours of the event:

- \* Any serious event (including on-site and off-site adverse events, injuries, side effects, deaths, or other problems) which in the opinion of the local investigator was unanticipated, involved risk to subjects or others, and was possibly related to the research procedures;
- \* Any serious accidental or unintentional change to the IRB-approved protocol that involves risk or has the potential to recur;
- \* Any publication in the literature, safety monitoring report, interim result or other finding that indicates an unexpected change to the risk/benefit ratio of the research;
- \* Any breach in confidentiality or compromise in data privacy related to the subject or others; or

7/10/2015

https://nugrant.unl.edu/era/orn/irb/viewPrintedMessage.php?ID=443291

\* Any complaint of a subject that indicates an unanticipated risk or that cannot be resolved by the research staff.

For projects which continue beyond one year from the starting date, the IRB will request continuing review and update of the research project. Your study will be due for continuing review as indicated above. The investigator must also advise the Board when this study is finished or discontinued by completing the enclosed Protocol Final Report form and returning it to the Institutional Review Board.

If you have any questions, please contact the IRB office at 472-6965.

Sincerely,

Julia Torquati, Ph.D. Chair for the IRB

Calia C. Tonquesti



NEBRASKA'S HEALTH SCIENCE CENTER

Office of Regulatory Affairs (ORA) Institutional Review Board (IRB)

June 29, 2015

Rachel Schmitz Sociology UNL - Via Courier

IRB # 482-15-XI

TITLE OF PROPOSAL: On the Street and On Campus: A Comparison of Life Course Trajectories among

Homeless and College Lesbian, Gay, Bisexual and Transgender Young Adults

**DATE OF REVIEW:** 06/29/2015

DATE OF FINAL UNMC ACCEPTANCE: 06/29/2015

The University of Nebraska Medical Center (UNMC) Institutional Review Board (IRB) for the Protection of Human Subjects has completed its review of the above-titled protocol. Please be advised that the UNMC IRB has accepted approval from the University of Nebraska - Lincoln (UNL) Institutional Review Board.

For the record, patients eligible for the study will be identified by UNMC and referred to UNL. It is understood that the UNL IRB is responsible for oversight of the above-titled research project including approval of any protocol amendments,

Finally, please be advised that acceptance by the UNMC IRB of the UNL IRB approval is valid for a period of five years from the initial date of review. If the study continues beyond the five year period, the project must be resubmitted in order to maintain an active status.

Sincerely,

Signed on: 2015-06-29 17:09:00.000

Gail Kotulak, BS, CIP IRB Administrator III Office of Regulatory Affairs

# Appendix B





DEPARTMENT OF SOCIOLOGY

711 Oldfather Hall Lincoln, NE 68588-0324 (402) 472-3631 FAX (402) 472-6070

#### PARTICIPANT INFORMED CONSENT FORM

"LGBT Young Adults Research Study"

You are invited to be in a research study exploring the experiences of lesbian, gay, bisexual and transgender (LGBT) young adults. You were selected as a possible participant because you identify as LGBT or another sexual or gender minority. Participants must be between 19 and 26 years old to participate. We ask that you read this form and ask any questions you may have before agreeing to be in the study.

My name is Rachel Schmitz and I am a doctoral candidate at the University of Nebraska-Lincoln. This study is being conducted for my dissertation research and its purpose is to understand LGBT young adults' experiences with their identity and their social relationships.

If you agree to be in this study, we would ask you to do the following things:

- Answer 5 screening questions to determine study eligibility.
- Participate in one face-to-face interview lasting approximately one hour.
- Be audiotaped during the interview.
- Be recontacted after the study is complete to review the findings if you choose to do so.

Upon completion of the face-to-face interview, you will receive \$20 cash for your participation in this study. The interview will take place in a private location agreed upon by the investigator and the participant, such as an office or reserved room at a public library.

The study has minimal risks for participants. You may be asked to recall memories that may cause mild psychological distress if you have had unpleasant experiences regarding your LGBT identity. Another risk of participating in this research is that there is a chance of your sexual or gender identity being disclosed if there were to be a breach in confidentiality. During the interview, I will ask you questions about your life while growing up, your family and peer relationships and how you came to identify as LGBT. You can skip any of these questions if you do not want to answer them, but if you don't complete the interview through to the end, you will not receive the \$20 cash.

Upon completion of the interview, you will be provided with a list of services and agencies available to young adults. Some of these services are free and others may have a fee depending on your ability to pay. Payment for psychological services is the responsibility of the participant.

Participants will benefit from the study by being afforded the opportunity to share their personal experiences. More broadly, participants will help us understand the unique experiences of LGBT young adults.



Everything you tell me will be confidential. The only people who will ever see your answers will only have an ID number; they will not know your name and your name will never be used in any written report. Data obtained from your responses will be encrypted and firewalled and stored on my personal computer that is password-protected. Only the project leader (PI) will access the data. Electronic and paper copies of the transcripts, the MP3 interview audio recordings uploaded from a digital recorder onto my personal computer and demographic questionnaires will be kept for a period of three years after data collection is complete. After five years, electronic copies of these materials will be deleted from my personal computer and all paper copies will be destroyed. All consent forms, demographic questionnaires and any paper copies of the transcripts will be stored in a locked file in the PI's office. If you choose to provide your information to be recontacted to review the findings, a master list with your contact information will be kept in a separate locked file.

#### Freedom to Withdraw:

Participation in this study is voluntary. You can refuse to participate or withdraw at any time without harming your relationship with the PI or the University of Nebraska-Lincoln. Your decision will not result in any loss of benefits to which you are otherwise entitled.

If you would like to be in this study after reading this consent form, please check the boxes below and sign on the line with your name or initials only, which implies you are giving your consent to participate. Please feel free to ask me any questions about the study before we begin.

If you have any questions about your rights as a research participant that have not been answered by the investigator or to report any concerns about the study, you may contact the University of Nebraska-Lincoln Institutional Review Board, telephone (402) 472-6965.

The primary researcher conducting this study is Rachel Schmitz. You may ask any questions you have now. If you have questions later, you are encouraged to contact her at raschm02@gmail.com or at 402-937-1734. The primary investigator's faculty advisor for this project, Dr. Kimberly Tyler, can be contacted at ktyler2@unl.edu and (402) 472-6073.

#### Consent, Right to Receive a Copy:

You are voluntarily making a decision whether or not to participate in this research study. Your checking the boxes below implies that you have decided to participate having read and understood the information presented. You will be given a copy of this consent form to keep.

Participant Initials:	Date:
Investigator Signature:	Date:
I agree to be audiotaped for the duration of the interview (chec	ck box if you agree)
I agree to be recontacted to review the findings once study is c (check box if you agree) Page 2 of 2	complete

# Appendix C

# Nebraska Lincoln

You are invited to be in a research study exploring lesbian, gay, bisexual and transgender (LGBT) college students' experiences. We are seeking LGBT college students between the ages of 19-26 to participate. You will receive \$20 in exchange for your time.

The goals of this study are to understand LGBT college students' experiences with their identity and their social relationships.

If you agree to be in this study, we would ask you to do the following things:

- Answer 5 screening questions to determine study eligibility.
- Participate in one face-to-face interview lasting approximately one hour.
- Be audiotaped during the interview.
- Be recontacted to review the findings if you so choose.

Please contact the primary investigator Rachel Schmitz (UNL) at raschm02@gmail.com or at 402-937-1734 (call/text) or the project advisor Dr. Kimberly Tyler at ktyler2@unl.edu or 402-472-6073 for more information regarding the study or if you are interested in participating. IRB Protocol Number: 15214

Thank you very much for your time.

| LGBT College Students |
|-----------------------|-----------------------|-----------------------|-----------------------|-----------------------|-----------------------|-----------------------|-----------------------|-----------------------|-----------------------|
| Rachel Schmitz        |
| raschm02@gmail.com    |
| 402-937-1734          | 402-937-1734          | 402-937-1734          | 402-937-1734          | 402-937-1734          | 402-937-1734          | 402-937-1734          | 402-937-1734          | 402-937-1734          | 402-937-1734          |
| LGB                   |
| Rac                   |
| rasc                  |
| 402                   | 402                   | 402                   | 402                   | 402                   | 402                   | 402                   | 402                   | 402                   | 402                   |

# Nebraska Lincoln

You are invited to be in a research study exploring lesbian, gay, bisexual and transgender (LGBT) homeless young adults' experiences. We are seeking LGBT young adults between the ages of 19-26 to participate who don't have a regular place to live. You will receive \$20 in exchange for your time.

The goals of this study are to understand LGBT homeless young adults' experiences with their identity and their social relationships.

If you agree to be in this study, we would ask you to do the following things:

- Answer 5 screening questions to determine study eligibility.
- Participate in one face-to-face interview lasting approximately one hour.
- Be audiotaped during the interview.
- Be recontacted to review the findings if you so choose.

Please contact the primary investigator Rachel Schmitz (UNL) at raschm02@gmail.com or at 402-937-1734 (call/text) or the project advisor Dr. Kimberly Tyler at ktyler2@unl.edu or 402-472-6073 for more information regarding the study or if you are interested in participating. IRB Protocol Number: 15214

Thank you very much for your time.

# Appendix D

Subject line: Young Adult Research Study

You are invited to be in a research study exploring the experiences of lesbian, gay, bisexual and transgender (LGBT) young adults. This study seeks individuals who identify as LGBT or another sexual or gender minority and currently don't have a regular place to live. Participants must be between 19 and 26 years old to participate. Please ask any questions you may have before agreeing to be in the study.

My name is Rachel Schmitz, a Ph.D. student at the University of Nebraska-Lincoln, and I am the primary investigator of this study collecting data for my dissertation. The purpose of this study is to understand LGBT homeless young adults' experiences with their identity and their social relationships. This study is entirely voluntary.

If you agree to be in this study, we would ask you to do the following things:

- Answer 5 screening questions to determine study eligibility.
- Participate in one face-to-face interview lasting approximately one hour.
- Be audiotaped during the interview.
- Be recontacted after the study is complete to review the findings if you choose to do so.

Upon completion of the in-person interview, you will receive \$20 cash for your participation in this study. The interview will take place in a private location agreed upon by the investigator and the participant, such as an office or reserved room at a library.

Please contact the primary investigator Rachel Schmitz at raschm02@gmail.com or 402-937-1734 (call or text) or the project advisor Dr. Kimberly Tyler at ktyler2@unl.edu or 402-472-6073 for more information regarding the study or if you are interested in participating.

Thank you very much for your time.

IRB Protocol Number: 15214

Subject line: Young Adult Research Study

University of Nebraska-Lincoln

You are invited to be in a research study exploring the experiences of lesbian, gay, bisexual and transgender (LGBT) college students. This study seeks individuals who identify as LGBT or another sexual or gender minority and are enrolled in college. Participants must be between 19 and 26 years old to participate. Please ask any questions you may have before agreeing to be in the study.

My name is Rachel Schmitz, a Ph.D. student at the University of Nebraska-Lincoln, and I am the primary investigator of this study collecting data for my dissertation. The purpose of this study is to understand LGBT college students' experiences with their identity and their social relationships. This study is entirely voluntary.

If you agree to be in this study, we would ask you to do the following things:

- Answer 5 screening questions to determine study eligibility.
- Participate in one face-to-face interview lasting approximately one hour.
- Be audiotaped during the interview.
- Be recontacted after the study is complete to review the findings if you choose to do so.

Upon completion of the in-person interview, you will receive \$20 cash for your participation in this study. The interview will take place in a private location agreed upon by the investigator and the participant, such as an office or reserved room at a library.

Please contact the primary investigator Rachel Schmitz at raschm02@gmail.com or 402-937-1734 (call or text) or the project advisor Dr. Kimberly Tyler at ktyler2@unl.edu or 402-472-6073 for more information regarding the study or if you are interested in participating.

Thank you very much for your time.

IRB Protocol Number: 15214

# Appendix E

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### AND

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at (402) 937-1734 for more information.

IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** 

for more information. IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### AND

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** for more information. IRB Protocol Number: 15214

### University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### <u>AND</u>

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** for more information. IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** 

for more information.
IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### AND

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time.

Please call/text Rachel at **(402) 937-1734**for more information.

IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### AND

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** for more information. IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

### AND

you are enrolled in college, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** for more information. IRB Protocol Number: 15214

### University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time.

Please call/text Rachel at **(402) 937-1734**for more information.

IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402)** 937-1734 for more information.

IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### AND

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time.

Please call/text Rachel at **(402) 937-1734**for more information.

IRB Protocol Number: 15214

### University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### AND

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** for more information. IRB Protocol Number: 15214

### University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time.

Please call/text Rachel at (402) 937-1734 for more information.

for more information. IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402)** 937-1734

for more information. IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time.

Please call/text Rachel at **(402) 937-1734**for more information.

IRB Protocol Number: 15214

# University of Nebraska - Lincoln Research Study

If you are between the ages of 19-26

#### **AND**

you don't have a regular place to live, and identify as lesbian, gay, bisexual or transgender, you may be eligible to participate & could receive \$20 in exchange for your time. Please call/text Rachel at **(402) 937-1734** 

for more information. IRB Protocol Number: 15214

# Appendix F

# List of Services for College Students at UNL and UNO

#### UNL

- Counseling and Psychological Services (CAPS) 402-472-7450
- Women's Center Counseling 402-472-9428
- LGBTOA+ Resource Center 402-472-1652
  - Consultation and assistance on LGBTQA+ issues; inclusive place to spend time and do homework; assistance with bias incidents on campus.
- LGBTQ At The U Support Group 402-472-7450
  - A confidential and safe group to support students who identify as lesbian, gay, bisexual, transgender, asexual, or may be questioning their sexual orientation. No registration required.
- Men at Nebraska Contact Felipe at wcmensprograms@unl.edu.
  - Join men, women, and people of all genders in discussing different forms of masculinity in our society and within our lives.

#### **UNO**

- **UNO Counseling Center** 402-554-2409
- Community Counseling Clinic 402-554-2727
  - Available services include, individual, group, marital and family work with children. Also, couples counseling related to personal, familial, crisis, special concerns/needs, professional, and career/vocational issues.
- Women's Resource Center 402-554-3884
- Gender and Sexuality Resource Center 402-554-2890; jhitchins@unomaha.edu (Director)

## List of Services for Homeless Young Adults in Lincoln and Omaha

#### Lincoln

#### CEDARS Street Outreach – 402-560-9356

- Provides street-based assistance to runaway, homeless, and at-risk youth in Lincoln, Nebraska. Without pressure or judgment, Street Outreach staff members support homeless youth as they work to leave the streets.

#### • The Bay – 402-310-5215

- A place for the kids of Lincoln to come and pursue their passions - whether that be skateboarding, art, music, or whatever else. The most important thing about THE BAY is helping kids reach their full potential.

#### • The Gathering Place – 402-476-7398

- Serves free evening meals to homeless and near-homeless individuals and families living in our community. Meals are served Monday through Friday from 5:00 p.m. to 6:00 p.m.

#### • Matt Talbot Kitchen and Outreach – 402-477-4116

- Devoted to serving homeless and near homeless men, women, and children, MTKO provides meals twice daily, everyday of the year. Outreach and advocacy programs include such services as on-site monthly health clinics, basic and emergency needs, life skills training, transitional housing, case management, nutrition counseling, and vision care

#### • **People's City Mission** – 402-475-1303

- Emergency shelter provides overnight shelter, three meals daily, shower and laundry facilities, case management, and referrals. The Family and Women's Shelter serves single women with or without children, married couples with children, and single fathers with children. The Men's Shelter serves single men. Guests staying longer than thirty days must be actively participating in a self-sufficiency program. Youth programs are offered for children ages 6 weeks and older staying in the shelter with their families.

#### • Fresh Start Home – 402-475-7777

- Transitional housing, basic and emergency needs, and supportive services for homeless women without children in their custody, 19 years or older, for 24 women at a time, for up to one year.

#### • Center Pointe – 402-474-4343

Adult and youth treatment programs offer outpatient treatment services for youth 13-18
and adults 19 and over with substance use disorders or co-occurring substance use and
mental health disorders; services are provided based on client need and many include
individual, group, and family counseling and psychiatric services; expected stay depends
on client need, but generally 6-8 months.

#### • The HUB – 402-471-8526

- Providing unconditional support and programming for young people, as they become productive, independent, active members of the community.

# $\bullet \quad \textbf{Bridge to Independence} - 402\text{-}471\text{-}9331$

- Bridge to Independence provides stable support for young people as they cross from foster care to adulthood.

#### **Omaha**

- National Runaway Safeline 1-800-RUNAWAY
- Youth Emergency Services (Y.E.S.) 402-345-5187
  - Serves homeless and at-risk youth by providing critically-needed resources which empower them to become self-sufficient. Helps youth in crisis get back on their feet in whatever way by meeting their immediate needs for food, shelter, clothing and safety, and creating a support system which helps them flourish.
- Jacob's Place Transitional Living 402-457-7000
  - Resource for youth ages 17-20 who struggle with a lack of housing, support, education and independent living skills because of multiple out-of-home placements.
- **Open Door Mission** 402-422-1111
  - Offers 816 safe, shelter beds to homeless men, women and children, serves over 2,000 hot, nutritious meals and provides preventive measures to more than 275 people living in poverty.
- Siena Francis House Homeless Shelter 402-341-1821
  - Shelter providing food, shelter, clothing and hope to homeless men, women, and children
- Stephen Center Emergency Shelter 402-731-0238
  - Works with homeless men, women and children.
- **Together** 402-345-8047
  - Offers a variety of services such as food pantry boxes, temporary transportation, ID and birth certificate services and short-term rent assistance.
- **Project Everlast** 402-384-4670
  - Promotes community resources to improve a foster youth's opportunities and networks for housing, transportation, health care.

# Appendix G

Screener Questions for Eligibility to Participate in LGBT Young Adults Research Study

1. What is your current age?(Must be between 19-26 to participate)
2. How do you currently define your gender?
Man
Woman
Transgender
Other:
3. Which of the following best describes your sexual orientation?  Gay Lesbian Bisexual Other:
4. Are you currently enrolled in college? Yes
No
5. Do you currently have a regular place to live? Yes No

# Appendix H

# LGBT Young Adult Demographic Questionnaire

This form will give me background information about you. Please let me know if you have questions or comments about any of these questions. You may skip any question that you do not feel comfortable answering.

1. How do you currently define your gender?
Man
Woman
Transgender
Additional details about your gender:
Other:
2. What is your race?
White or Caucasian
African-American or Black
Latino/Latina or Hispanic
Asian-American or Asian Pacific Islander
Native American or American Indian
Bi- or Multi-racial:
Other:
3. What is your current age?
4. Which of the following best describes the area you in which you live?
Urban
Rural
Suburban
5. Which of the following best describes your sexual orientation?
Gay
Lesbian
Bisexual
Other:
6. How long have you identified as that sexual orientation?
7. What would you say best describes your family class status while you were growing up
Working class
Middle class
Upper-middle class
Upper class
Other:

8. What is the highest level of your parent or parent's education?
Some high school
Graduated high school
Some college
Associate's degree
Bachelor's degree
Professional or advanced degree
Other:
Don't know/unsure
9. Did your family ever receive any public assistance, such as welfare, Aid to Families with Dependent Children (AFDC) or food stamps (SNAP) while you were growing up?  Yes  No
Unsure
Unsure
10. Did you and your family ever live in public housing or a housing project when you were growing up?  Yes  No  Unsure
Unsure
11. What would you say best describes your relationship status?  Married In a domestic partnership In a relationship Single Other:
12. How did you hear about this research project?
13. Is there anything else that you would like to share about yourself?

# Appendix I

### **Interview Questions for LGBT Young Adults**

- 1. Please describe your relationships with your parents while you were growing up.
- 2. Please describe your relationships with your siblings and other family members while you were growing up.
- 3. How did you come to identify as (insert LGBT identity here)?
- 4. During what time of your life did you come to identify as (insert LGBT identity here)?
- 5. If you have come out, how did your family react to you coming out as LGBT?
- 6. If you have come out, how did your friends react to you coming out as LGBT?
- 7. If you have not come out, can you tell me why you are making this choice? How does it affect your social interactions?
- 8. How would you describe your experiences while in middle school and high school? (i.e. with family, friends, peers, teachers, school officials).
- 9. Can you please describe your current family relationships (i.e. support, conflict, etc.).
- 10. Can you please describe your current relationships with friends and peers.
- 11. How do you feel your sexual orientation/gender identity has shaped your relationships or ability to form relationships? Can you give me some examples?
- 12. Please describe your experiences as an LGBT young adult.
- 13. What challenges have you faced being LGBT? How does being (insert a student or homeless) affect these challenges related to your identity?
- 14. How have you responded to these challenges?
- 15. What do you see as the best thing(s) about being LGBT?
- 16. How would you describe the community acceptance towards LGBT young adults/people?
- 17. What do you see in your future?
- 18. What advice would you give to other LGBT young adults?
- 19. Is there anything else you would like to add that we didn't cover in the interview or that you would like to add to your responses?
- 20. Can you please tell me why you wanted to participate in the study and be interviewed?